

# WAGE DYNAMICS NETWORK

# WORKING PAPER SERIES NO 1309 / MARCH 2011

SKILLS AND WAGE INEQUALITY IN GREECE EVIDENCE FROM MATCHED EMPLOYER-EMPLOYEE DATA, 1995-2002

by Rebekka Christopoulou and Theodora Kosma



# WORKING PAPER SERIES

NO 1309 / MARCH 2011

WAGE DYNAMICS NETWORK

# SKILLS AND WAGE INEQUALITY IN GREECE

# EVIDENCE FROM MATCHED EMPLOYER-EMPLOYEE DATA, 1995-2002 <sup>1</sup>

by Rebekka Christopoulou<sup>2</sup> and Theodora Kosma<sup>3</sup>

NOTE: This Working Paper should not be reported as representing the views of the European Central Bank (ECB). The views expressed are those of the authors and do not necessarily reflect those of the ECB.







This paper can be downloaded without charge from http://www.ecb.europa.eu or from the Social Science Research Network electronic library at http://ssrn.com/abstract\_id=1773210.

1 The work was conducted within the framework of the Eurosystem Wage Dynamics Network (WDN). We thank for their helpful comments Heather Gibson, Stephen Hall, Juan F. Jimeno, Ana Lamo, Theodoros Mitrakos, Vassilis Monastiriotis, Lia Papapetrou, Frank Smets, the participants of the WDN meetings and an anonymous referee of the ECB Working Paper Series. Finally, we would like to thank the National Statistical Service of Greece for providing access to the Structure of Earning Survey data to the Bank of Greece. Rebekka Christopoulou acknowledges financial support from the Greek Ministry of Economy and Finance. The views expressed in the paper are of the authors and do not necessarily reflect those of the Bank of Greece.

2 Bronfenbrenner Life Course Center, Cornell University, Ithaca, NY 14853, USA; e-mail: rc448@cornell.edu.

Economic Research Department, Bank of Greece, 21 E. Venizelos Ave.,

GR-10250 Athens, Greece; e-mail: tkosma@bankofgreece.gr.

#### Wage Dynamics Network

This paper contains research conducted within the Wage Dynamics Network (WDN). The WDN is a research network consisting of economists from the European Central Bank (ECB) and the national central banks (NCBs) of the EU countries. The WDN aims at studying in depth the features and sources of wage and labour cost dynamics and their implications for monetary policy. The specific objectives of the network are: i) identifying the sources and features of wage and labour cost dynamics that are most relevant for monetary policy and ii) clarifying the relationship between wages, labour costs and prices both at the firm and macro-economic level.

The WDN is chaired by Frank Smets (ECB). Giuseppe Bertola (Università di Torino) and Julian Messina (Universitat de Girona) act as external consultants and Ana Lamo (ECB) as Secretary.

The refereeing process of this paper has been co-ordinated by a team composed of Gabriel Fagan (ECB, chairperson), Philip Vermeulen (ECB), Giuseppe Bertola, Julian Messina, Jan Babecký (CNB), Hervé Le Bihan (Banque de France) and Thomas Mathä (Banque centrale du Luxembourg).

The paper is released in order to make the results of WDN research generally available, in preliminary form, to encourage comments and suggestions prior to final publication. The views expressed in the paper are the author's own and do not necessarily reflect those of the ESCB.

#### © European Central Bank, 2011

#### Address

Kaiserstrasse 29 60311 Frankfurt am Main, Germany

Postal address Postfach 16 03 19 60066 Frankfurt am Main, Germany

**Telephone** +49 69 1344 0

Internet

http://www.ecb.europa.eu

Fax +49 69 1344 6000

All rights reserved.

Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the authors.

Information on all of the papers published in the ECB Working Paper Series can be found on the ECB's website, http://www. ecb.europa.eu/pub/scientific/wps/date/ html/index.en.html

ISSN 1725-2806 (online)

# CONTENTS

Abstract				
No	n-tec	hnical summary	5	
1	Intro	oduction	7	
2	2.1	a and timing The sample The timing: an overview of the Greek economy between 1995-2002	10 10	
3	Obs	erved wage changes	15	
4	The	methodology	17	
5	Resi	lts The added value of controlling	19	
	5.1	for employer heterogeneity	20	
	5.2	Aggregate price and composition effects	24	
	5.3	Wage changes due to skill	25	
	5.4	Economic developments versus		
		predetermined changes	32	
6	Con	clusion	34	
Re	feren	ces	36	
Appendix				

#### Abstract

This paper examines changes in the Greek wage distribution over 1995-2002 and the role of skills in these changes using a matched employer-employee data set. This data set enables us to account for firm heterogeneity and obtain a more refined picture of the impact of skills. The methodology adopted is the Machado-Mata decomposition technique, which separates the part of wage changes that is due to changes in the job/employer and employee characteristics from the part due to changes in the returns to these characteristics. Our results indicate that the role of skills has been decisive. The skill return effects in combination with the composition effects of tenure, which are arguably responsive to economic developments and market conditions, have had an important contribution to the changes in the Greek wage distribution. On the other hand, the impact of predetermined demographic changes, as those captured by the age and education composition effects, has been relatively milder.

JEL classification: J31

Keywords: Returns to skill; Wage inequality; Quantile regression

### Non-technical summary

In this paper we examine how the distribution of individuals' wages has changed in Greece, over the period 1995-2002, and what has been the role of skills in these changes. Following topical trends in the literature that point to the importance of other personal characteristics beyond education in shaping wages, we focus on the analysis of a broader dimension of employees' skills. We consider the contribution of education and age-potential labour market experience that enhance ones competence irrespective of the employer and also of tenure that boosts the accumulation of firmspecific human capital.

In our analysis we use a matched employer-employee data set and separate, at each decile of the wage distribution, the part of wage changes that is due to changes in the job/employer and employee characteristics from the part that is due to changes in the returns to these characteristics. Benefiting from the matched employer-employee data set we analyze the contribution of personal skills to changes in the wage distribution controlling for firm characteristics. By accounting for firm heterogeneity we are able to gain a better insight into the impact of worker characteristics.

We find that over 1995-2002 wage inequality increased; more so for men and those at the upper end of the wage distribution. The contribution of the composition and return effects of skills to the evolution of Greek wage dynamics has been important. Of the skill composition effects tenure and education effects have had the most significant impact. The composition effects of tenure are mostly negative and by reducing wages more for those in the lower part of the wage distribution, contribute to the observed pattern of wage inequality. On the hand, the composition effects of education are positive and also contribute to higher wage inequality. This result contradicts standard expectations for a negative relationship between rising education and wage equality and it is possibly the outcome of higher wage dispersion among the high skilled.

Our analysis of the price effects of skills suggests that, even though, they are smaller in magnitude than the respective skill composition effects they have also had an important role in shaping the changes in the Greek wage distribution. The price effects of skills have formed a U-shaped pattern along the wage distribution, which, though not a proof of, can be considered as being broadly in line with the routinization hypothesis; a variant of the of skill-biased technical change theory. As to the sign of individual price effects, the age price effects are negative while the tenure and education return effects are positive. Interestingly, the U-shaped pattern is more or less sustained for all individual skill price effects. The sign and U-shaped pattern of the age price effects could be regarded as partly reflecting the interaction between institutional changes that are not age neutral and different unobserved ability of employees along the wage distribution.

Our results imply that economic developments have had a significant contribution to the changes in the Greek wage distribution. Indeed, we find that the skill return effects in combination with the composition effects of tenure, which are arguably responsive to economic developments, have had a significant contribution. On the other hand, predetermined demographic changes, as those captured by the composition effects of age and education, have had a relatively milder impact.

# 1. Introduction

Education and job experience are traditionally seen as the most important dimensions of 'skill' for labour market participants. Highly educated and experienced workers have many job opportunities, receive high wages and enjoy good working terms and conditions. Through recent decades a variety of labour market outcomes has been linked to education in countries with different degrees of institutional flexibility. For instance, the rising wage inequalities in the 'Liberal' Anglo-Saxon countries have been attributed to the increased demand for skilled workers. Further, in the less flexible continental European countries, the relative increase in the unemployment of unskilled workers has also been attributed to the increased demand for skills (Krugman 1994; Blanchard and Wolfers 2000).

Since the early 90s, the skill-biased technical change (SBTC) hypothesis has been the prevalent explanation for the growing wage inequalities (Katz and Autor 1999; Acemoglu 2002). Briefly, SBTC assumes that technology biases labour demand in favour of the skilled and against the unskilled. Lately, however, the SBTC has been losing ground over the novel idea of 'routinization'. The 'routinization' hypothesis is a modified version of the SBTC hypothesis that takes the focus away from education and experience and moves it to the type of job content and the degree to which it can be routinized i.e., substituted by machinery and/or computers. It essentially assumes that technology increases demand for both high-skilled and low-skilled workers and decreases the demand for middle-skilled workers, as it replaces human labour mainly in routine tasks typically entailed in middle-skilled jobs (Autor, Levy and Murnane 2003; Autor, Katz and Kearny 2006, 2008; Goos and Manning 2007; Goos, Manning and Salomons 2009). While STBC and the variants of it have dominated Labour Economics a concurrent strand of the literature challenges the idea that technical change is the main driver of wage inequalities. This literature argues that inequalities by skill are mostly driven by non-market, mainly socio-demographic changes and has justly earned itself the title of the 'revisionist' (Card and DiNardo 2002; Lemieux 2006).

Along these research dynamics and since the early 1990s the effect of education on inequality and labour market outcomes in Greece has attracted the attention of researchers. Most of these studies analyse the relationship between skills and pay and focus mainly on the education dimension of skill (Tsakloglou and Cholezas 2005 give a brief review). The bulk of the research utilizes data from the Greek Household Budget Survey that provide information on consumption expenditures, incomes and socio-economic characteristics of the households and their members. This information has allowed the estimation of the returns to skills in the Greek labour market from the mid-1970s till the late 1990s in Mincer-type wage equations. The available evidence suggests that both overall wage inequality and the returns to education declined between the mid-1970s and the 1980s, but recovered again during the 1990s. This pattern has been attributed to interactions between an expanding educational system, a stagnant demand for educated workers and changing institutional structures.

As in many other countries, the returns to education seem to have evolved differently across the wage distribution; the results are however mixed. In particular, by estimating Mincer wage equations on data from the early 1990s using quantile regressions, Martins and Pereira (2004) find that Greece is the only country out of 16 analyzed that shows higher returns to education at the lower end of the wage distribution. Conversely, Cholezas (2004) examines Greek wages for the years 1974, 1988, 1994 and 1999, and finds that in most cases returns to education follow a U-shaped pattern across the wage distribution.

In this paper we use a newly developed matched employer-employee data set, covering a recent period of time (1995-2002), to examine how the distribution of individuals' wages has changed in Greece and what has been the contribution of skills to these changes. First, following topical trends in the literature that point to the importance of other personal characteristics beyond education in shaping wages, we focus on the analysis of a broader dimension of employees' skills. In particular, we consider the contribution of education and potential labour market experience that enhance ones competence irrespective of the employer and also of tenure that boosts the accumulation of firm-specific human capital.<sup>1</sup> Second, benefiting from the matched employer-employee data set we analyze the contribution of personal skills to changes in the wage distribution controlling for firm characteristics; by accounting for firm heterogeneity we are able to gain a better insight into the 'true' impact of worker

<sup>&</sup>lt;sup>1</sup> The relationship between wages, labour market experience and tenure has attracted the attention of many studies (e.g. Dustmann and Meghir, 2005).

characteristics.<sup>2</sup> Third, we employ a methodology of counterfactual decomposition, recently developed by Machado & Mata (2005). This methodology moves beyond the simple estimation of returns to skills; it allows us to separate, at each decile of the wage distribution, the part of wage changes that is attributable to compositional changes of individual and workplace characteristics (net composition effects) from the part that is attributable to changes in the returns to these characteristics (net price effects). In interpreting these decomposition results we are able examine the relative importance of economic developments and predetermined compositional changes (e.g. changes in age, educational attainment) in shaping the changes in the wage distribution in Greece. Traditionally, economic developments are assumed to influence returns. However, economic developments can also influence certain employee (i.e. tenure) and workplace characteristics. Given that we can perform the decomposition for the return and composition effects of each skill separately, economic developments can now be related not only to skill return effects but also to skill composition effects (cf. Christopoulou et. al. 2010).

We find that over 1995-2002 wage inequality increased; more so for men and those at the upper end of the wage distribution. The decomposition results show that the role of skills has been essential in shaping the changes in the Greek wage distribution. Specifically, the combined impact of the skill price effects and the composition effects of tenure, which are very likely responsive to economic developments, has been significant. <sup>3</sup> On the other hand, predetermined demographic changes as those captured by the composition effects of education and age have had a relatively milder impact.

The paper is structured as follows: Section 2 describes the data and provides a description of the economic environment in Greece the period under investigation. Section 3 provides a descriptive analysis of wage changes, section 4 presents the methodology and section 5 discusses the results. The discussion of the results is given in four stages; we comment on the value added of controlling for employer heterogeneity in section 5.1, in 5.2 we give a general introduction to the decomposition results, in 5.3 we focus on the skill effects, and in 5.4 we isolate the

<sup>&</sup>lt;sup>2</sup> Arguably, omission of firm variables will lead to unbiased estimates of worker returns only if firm and worker characteristics are uncorrelated; but this is very unlikely to hold in practise (cf. Hamermesh, 2008).

<sup>&</sup>lt;sup>3</sup> Economic developments are assumed to encompass market conditions, i.e. demand and supply considerations and also changes in the institutional environment.

predetermined skill effects from the part that is induced by economic developments. A final section concludes.

### 2. Data and Timing

The data used in the empirical analysis are obtained from the Greek Structure of Earnings Survey (SES), which is compiled by the National Statistical Service of Greece. The Structure of Earnings Survey was first conducted in 1995 in the EU member states with the aim of compiling a dataset comparable across countries. This dataset would then serve as a useful basis for analysing the progress of economic and social cohesion. The survey was again conducted in 2002 and it has been decided that the survey will be repeated every four years.<sup>4</sup>

The SES contains rich information on the structure and distribution of earnings and characteristics of employers and employees for two years: 1995 and 2002. Therefore, in comparison to household surveys that have been used in the literature to date, the SES has two important advantages. First of all, it avoids the measurement error problems of the household surveys.<sup>5</sup> Further, as already emphasized above, it enables controlling for both workers' and firms' characteristics when estimating wage equations. Its timing is also advantageous: it offers a more recent view of the labour market in comparison to previous studies and it coincides with a period of interesting economic developments. Next, the sample and its timing are discussed in turn.

#### 2.1. The sample

The sample of the Structure of Earning Survey is constructed by threedimensional stratified random sampling covering firms of more than 10 employees in sectors such as manufacturing, construction and services (NACE C-K). The process of deriving the sample is the following: in the first step a sample of firms from the firm registry is selected, in the second step the sample of the local units belonging to the firms of the first stage is selected, and in the final step a sample of employees

<sup>&</sup>lt;sup>4</sup> More details on the aim of the Structure of Earning Survey can be found on the website of the National Statistical Service of Greece (<u>www.statistics.gr</u>).

<sup>&</sup>lt;sup>5</sup> It is widely documented in the literature that household surveys are contaminated with a significant degree of measurement error. The data on wages/income are mostly affected by this measurement error; individuals do not exactly recall their income and pay components or, for various reasons, are not willing to provide accurate information on their income sources.

belonging to the local unit is selected. Before the selection, firms are classified into strata according to region, economic activity (NACE 2-digit) and firm size (defined by the number of employees in the firm).

The data available for the employees contain information on gender, age, the education level completed, tenure with the current employer. The data on job characteristics refer to the type of contract (part-time or full-time, contract of definite or indefinite length), the occupation, and whether the job entails supervisory duties. The data on employer characteristics contain information on the firm size, industry, location, main market in which the product of the firm is sold (regional, national, European or global), and the type of collective agreement enforced in the firm (national, sectoral, or firm level agreement).

The Structure of Earnings Survey also contains detailed information on the gross monthly earning of the employee, the various pay components such as overtime, irregular bonuses, hours worked and overtime hours. From the information provided we create the variable referring to hourly earnings including overtime and regular bonuses, which we use in the econometric analysis. More precisely, we use real hourly earnings (deflated by the Harmonized CPI).

Before the econometric analysis we subject the data to a thorough 'cleaning'. Incomplete or inaccurate observations are unavoidably deleted. Employees with age 15 to 65 are included; employees with earnings below the 1st and above the 99th percentiles are excluded.<sup>6</sup> After the data inspection and cleaning we end up with 38701 observations for 1995 and 41449 for 2002.

Table 1 provides selected information on the final 'clean' version of the sample. Firstly, following the widely-documented worldwide trend, the proportion of female employees has increased. Secondly, the average years of education have increased. Thirdly, average tenure with the same employer has decreased. This might be explained by a series of developments in the Greek economy. Specifically, there has been an increase in the proportion of employees under contracts of definite length.<sup>7</sup>

<sup>&</sup>lt;sup>6</sup> Our analysis requires a homogeneous sample between the two cross sections. We thus have to account for some differences between the two waves of the SES. For instance, observations for sectors covered only in one wave were deleted.

<sup>&</sup>lt;sup>7</sup> This is verified by the sample; the proportion of employees not having contracts of indefinite length has increased from around 2% in 1995 to 8.8% in 2002. Given the limited variability in the variable that captures alternative contract types in the 1995 wave we do not control for contract types in our regression analysis.

Also there has been an increase in newcomers in the labour force, possibly driven by the increase in the working age population. This development is also observed in our sample, as the proportion of young workers increases in the 2002 wave. Moreover, there has been a process of integrating immigrants in the Greek labour market (the SES has also started including them in the sample).<sup>8,9</sup>

	_			
Employee charac	teristics	1995	2002	Change
Female (%)		31.70	37.36	5.66
Years of educatio	on (average)	10.57	11.49	0.92
Years of tenure (a	average)	10.08	8.26	-1.82
Age:	15-24 years (%)	5.92	7.30	1.38
	25-34 years (%)	29.95	32.97	3.02
	35-44 years (%)	34.09	30.26	-3.83
	45-64 years (%)	30.01	29.44	-0.57
Employer charact	teristics			
Private ownership	o (%)	69.61	83.50	13.89
Firm size:	10-19 employees (%)	9.74	12.35	2.61
	20-49 employees (%)	21.90	16.71	-5.19
	50-99 employees (%)	21.17	10.88	-10.29
	>100 employees (%)	47.19	60.05	12.86
Manufacturing se	ector (%)	48.30	36.13	-12.17

Table 1: Sample characteristics

Note: % refers to % of employees in the sample.

Immigrant workers are likely to have more job mobility and also work under contracts of definite length.

There are some changes in firm characteristics that are also worth mentioning. The proportion of employees working in the private sector has increased, and so has

<sup>&</sup>lt;sup>8</sup> For a detailed analysis of Greek labour market developments between 1995 and 1999 see Sabethai

<sup>(2000).</sup> <sup>9</sup> Controls for immigrant status are not included in our analysis due to insufficient data especially in the 1995 wave.

the proportion of employees working in bigger firms (with more than 100 employees). The former fact may be related to the process of privatizations. Finally, the manufacturing employment seems to have followed a decreasing trend the period under investigation.

#### 2.2. The timing: an overview of the Greek economy between 1995-2002

Table 2 below presents various indicators that provide a general picture of the economic environment in Greece over the period 1995-2002. This period was a special period for Greece as it coincided with the years preceding the euro adoption and the need to fulfil the accession criteria. As one can see in the table, Greece 'delivered' in terms of macroeconomic performance; it experienced a high and increasing GDP growth rate (at the same time that the growth rate in two of the EU's core countries, Germany and France, as well as in the US was low and decreasing). Following the requirement for the euro adoption, inflation was also significantly reduced, with the increase in the unemployment rate being a possible consequence of policies aiming at this reduction.

In addition, the period 1995-2002 was also characterised by significant demographic, macroeconomic and institutional changes - some country-specific and some common across advanced countries. For the demographic developments, we have already gotten a flavour from the description of the sample characteristics. The Greek labour market has experienced an increase in female labour force participation, like the majority of the OECD countries. Additionally, the inflow of immigrants has also increased substantially.<sup>10</sup>

<sup>&</sup>lt;sup>10</sup> This is documented in detail by Zografakis, Kontis and Mitrakos (2009).

	1995	2002	Change
Real GDP growth (2000 constant prices)	2.1	3.9	1.8
Unemployment rate	9	10.3	1.3
Inflation rate	9.8	3.7	-6.1
Proportion of foreign population	2.8 (1998)	4.1	1.3
Female labour force participation	44.3	50.1	5.8
Population share of 15-24 year-olds	20.4	19.3	-1.1
Population share of 25-49 year-olds	51.3	54.4	3.1
Population share of 50-64 year-olds	28.3	26.3	-2
Trade in goods and services to GDP	18.8	23.3	4.5
Share of ICT investment in total gross fixed capital formation	10	11.5	1.5
Minimum relative to median wages of full-time workers	0.53	0.46	-0.07
Strictness of employment protection legislation (range 0-6)	3.5	3.5	0
Overall product market regulation (range 0-4)	2.8 (1998)	2	-0.8

Table 2: Macroeconomic background, demographics, and market regulation

Source: OECD Statistics

At the same time, as indicated by the measures of investment in Information and Communication Technologies (ICT), technical change was gradually transforming parts of the production process; openness was also increasing as the evolution of the ratio of trade-to-GDP shows. Coming now to institutional flexibility, there are indications that product market regulation was decreasing.<sup>11</sup> Employment protection legislation, on the other hand, seems to have been unchanged the period under investigation. However, despite the unchanged employment protection legislation index and the fact that the Greek system of wage bargaining is still considered to be broadly regulated (for more details see DuCaju *et.al.* 2008) the Greek labour markets were also becoming more flexible. <sup>12</sup> This flexibility involved the introduction of flexible working time arrangements (i.e. part-time jobs) and more flexible types of

<sup>&</sup>lt;sup>11</sup> The product market regulation indicator presented in Table 2 captures mainly barriers to entry. For a detailed description of its construction see Conway et. al (2005).

<sup>&</sup>lt;sup>12</sup> The perceptions and experiences of market participants may also be indicative of the 'effective' degree of employment protection the period under investigation. Specifically, evidence from EU ad hoc surveys conducted in 1999, indicates that firms do not consider the Greek labour markets to be as inflexible as they may initially seem. More precisely, Greek firms in the industry do not consider regulation as an obstacle for adjusting employment and production during demand fluctuations (for a detailed description see Sabethai, 2000 and references therein).

employment contracts such as contracts of definite length, contracts for the provision of specific services and for carrying out a specific task (for a more detailed analysis see Sabethai, 2000). <sup>13</sup>

Overall, the period under investigation is a period of significant economic, institutional and demographic developments. We now proceed to analyse whether the distribution of wages has changed in Greece over 1995-2002 and which are the factors that have mostly contributed to these changes.

#### **3.** Observed wage changes

The direction, magnitude and nature of wage changes between the two sample waves is roughly indicated by changes in the measures of mean and standard deviation, which amount to 0.052 and 0.087 log points respectively, when taking men and women together. These numbers reflect a pattern of slow and asymmetrical wage movement. Still, it says little if not compared with similar changes in other countries. Christopoulou, Jimeno and Lamo (2010) provide a comparison of the two measures between Greece and eight other EU countries, for which comparable SES data is available. They show that putting Ireland's and Hungary's impressive wage growth aside, Greece's average wage change is well in line with the experiences of the other European countries. The change in the standard deviation of hourly wage, though, stands out. After Germany, Greece is the second country in the group with the biggest increase in wage dispersion (for more details see Christopoulou, Jimeno and Lamo, 2010, Table A1, p. 32).

Figure 1 and Table 3 describe the shift in the Greek wage distribution in detail, serving to reveal a very interesting picture. In the aggregate sample, real hourly wages have remained more or less constant up to the 5th decile of the distribution and have monotonically increased thereafter.

<sup>&</sup>lt;sup>13</sup> Undoubtedly, labour market outcomes are not exogenous to economic developments, such as increased trade openness and product market deregulation; thus the structure of labour market institutions may evolve further in the coming years. In particular, the literature predicts that competition from low cost producers and product markets deregulation may indirectly have an impact on trade unions' power and employment regulations (for a discussion on the effects of product market regulation see Fiori et. al. 2008).

In other words, the moderate increase in mean wages has not been shared equally among employees, but rather those at the upper part of the wage distribution (high wage workers) have become better off while the wages of those at the lower part of the distribution have not increased as much. As a result wage inequality has increased and this can be attributed to increasing upper tail inequality; on the other hand, lower tail inequality has not increased as much (Table 3). Christopoulou, Jimeno and Lamo (2010) show that Germany and the Netherlands have experienced comparable increases in overall wage inequality; however, in these two countries inequality has increased more for those at the lower end of the wage distribution. In fact, the Greek experience appears to be similar to that of the US and the UK, over the same eight year period (Autor, Katz and Kearny 2006; Autor *et. al.* 2008 and Machin and Van Reenen 2007).

Figure 1: Real hourly wages by decile



When disaggregating the sample by gender, the picture becomes slightly different. Specifically, the wage movement is still concentrated at the upper part of the distribution both for men and women. For men, however, at the bottom of the distribution one can clearly see some wage falls, which wear out when moving towards the middle, switching to wage increases after the 5th decile (see Figure 1). In contrast, the wages of women have not decreased at all. Instead, starting from a low relative wage level in 1995, women have been catching up, experiencing wage

increases from the 2nd decile of the distribution onwards. As a result, the overall increase in wage inequality has been larger for men (see Table 3).

These different experiences motivate our choice to analyse wage changes both for the full sample (pooled regressions) and separately by gender. In the first case we assume that returns to characteristics are the same for both men and women and there are only intercept differences captured by a gender dummy. In the second case we allow for full heterogeneity in wage determination by estimating separate regressions for males and females.

	Table 3: Key indicators of the wage distribution										
		Std. Dev	Median	P90/P10	P50/P10	P90/P50	Gini coef.				
All	1995	0.38	1.88	1.69	1.30	1.30	0.22				
	2002	0.47	1.89	1.85	1.33	1.40	0.27				
	Change	0.09	0.01	0.16	0.02	0.10	0.05				
Males	1995	0.38	1.98	1.67	1.32	1.27	0.22				
	2002	0.48	2.01	1.86	1.37	1.36	0.28				
	Change	0.10	0.03	0.20	0.06	0.09	0.06				
Females	1995	0.32	1.67	1.59	1.21	1.31	0.19				
	2002	0.41	1.73	1.74	1.25	1.39	0.24				
	Change	0.09	0.06	0.15	0.04	0.08	0.05				

### 4. The Methodology

The analysis relies on the estimation of extended Mincer equations for log (real) hourly wages at different deciles of the wage distribution for each year t, using the quantile regressions method:

$$\ln w_{it}^{\mathscr{G}} = a_t^{\mathscr{G}} + \sum_j \beta_{jt}^{\mathscr{G}} X_{jit}^{\mathscr{G}} + \varepsilon_{it}^{\mathscr{G}}, \quad with \ Q_{\mathscr{G}}(\ln w \mid X) = \beta^{\mathscr{G}} X^{\mathscr{G}}$$

where  $w_i$  represents the wage of individual *i*, *X* is the vector of observable labour market characteristics,  $a^{\partial}$  is a constant, and  $\beta^{\mathcal{G}}$  is the vector of parameters.  $Q_{\mathcal{G}}(lnw/X)$ denotes the  $\mathcal{G}$ th conditional quantile of lnw given *X*.  $\varepsilon$  is the stochastic error. Given these estimates, we decompose the change between the 1995 and 2002 log wage distributions into a part that is due to changes in labour market characteristics and a part that is due to changes in the returns to these characteristics. The decomposition by decile is performed using the quantile decomposition approach recently developed by Machado and Mata (2005).<sup>14</sup> This decomposition method can be considered as a generalization of the Oaxaca and Blinder (1973) technique; the latter by focusing exclusively on the mean of the wage distribution could not be informative in the analysis of wage inequality. The decomposition is performed on the basis of the following idea. Two counterfactual densities are estimated: (i) the wage density corresponding to the 1995 distribution of characteristics with returns held constant at 2002 levels, and (ii) the wage density corresponding to the 2002 distribution of characteristics with returns held constant at 1995 levels.

The linearity of the quantile regression implies:

$$\ln w_{02}^{g} - \ln w_{95}^{g} = \left(a_{02}^{g} - a_{95}^{g}\right) + \sum_{j} \beta_{j02}^{g} \left(\overline{X}_{j02}^{g} - \overline{X}_{j95}^{g}\right) + \sum_{j} \left(\beta_{j02}^{g} - \beta_{j95}^{g}\right) \overline{X}_{j95}^{g} + \left(\overline{\varepsilon}_{02}^{g} - \overline{\varepsilon}_{95}^{g}\right)$$

where  $w_t^{\mathfrak{g}}$  is the  $\mathfrak{G}th$  decile of the wage distribution in year t,  $\overline{X}_{jt}^{\mathfrak{g}}$  is the vector of mean characteristics of decile  $\mathfrak{g}$  at year t, and  $\overline{\varepsilon}_t^{\mathfrak{g}}$  is the mean of the unobserved component.

We carry out the computation of mean characteristics by decile according to the adaptation of the Machado-Mata bootstrap method by Albrecht, Bjorklund and Vroman (2003). To describe it in simple terms, for each year, we draw a random subsample of 100 observations (i.e. individuals) from the whole sample. We sort the observations of the sub-sample by hourly pay and obtain the resulting decile values of the variables of interest. We repeat these steps 500 times, obtaining 500 values per

<sup>&</sup>lt;sup>14</sup> Various methodologies have been developed to analyze changes in the distribution of wages. One methodology that has been widely used in the literature is the one developed by DiNardo *et.al* (1996). This technique, however, does not explicitly model the role of prices (see Autor *et. al* 2005). The Machado and Mata methodology by being parametric can decompose the changes in the wage distribution into the part that is due to changes in the distribution of covariates and the part that is due to changes in the distribution of covariates and the part that is due to changes in the coefficients (for a discussion see Machado and Mata, 2005). Another quantile based decomposition technique, which is in the spirit of the Machado and Mata methodology, has recently been developed by Melly (2005 and 2006). This technique is numerically identical to that of Machado and Mata as the number of simulations goes to infinity (see Melly, 2006). It is a useful alternative to the Machado Mata method if one is interested in analysing the changes in the distribution of wages that are also due to changes in the residuals. The latter effects are, though, beyond the scope of the present study.

variable in each decile. We then calculate the average of these 500 values in each decile, ending up with 10 values per variable (i.e. one for each decile).

Once the mean characteristics have been calculated, the wage changes by decile over the period 1995-2002 are decomposed as follows:  $\left(a_{02}^{g} - a_{95}^{g}\right)$  is due to changes in unobserved features common among employees and due to changes in the reference categories (dummies);  $\sum_{j} \beta_{j02}^{g} \left(\overline{X}_{j02}^{g} - \overline{X}_{j95}^{g}\right)$  is due to changes in (employer or employee) observable characteristics net of any price effects (composition effect);  $\sum_{j} \left(\beta_{j02}^{g} - \beta_{j95}^{g}\right)\overline{X}_{j95}^{g}$  is due to changes in the returns to (employer or employee) characteristics net of any composition effects (price effects); and  $\left(\overline{\varepsilon}_{02}^{g} - \overline{\varepsilon}_{95}^{g}\right)$  is due to changes in the remaining unobserved component.

Customarily in the empirical analysis of Mincer equations X only includes variables representing individual/employee characteristics (i.e. educational level, age, age squared, tenure, tenure squared, a constant, and a gender dummy). An important reason for this is that the arrival of matched employer-employee datasets has been relatively recent. In a sweeping review of the international literature, Abowd and Kramarz (1999) note that virtually all papers using matched employer-employee data appeared after the late 1990s and, in their majority, the databases used have been European. Taking advantage of the information available in the Greek Structure of Earnings Survey we introduce variables that control for job and employer characteristics when analyzing the contribution of skills to changes in the wage distribution.

# 5. Results

In this section we first estimate Mincer equations in two alternative specifications; one including only employee characteristics -specification 1- and one including employee as well as employer and job characteristics - specification 2 - (subscripts t and i are suppressed for simplicity):

$$\ln w^{\vartheta} = a^{\vartheta} + \sum_{j} \beta_{j}^{\vartheta} X_{j}^{\vartheta} + \varepsilon^{\vartheta}$$
<sup>(1)</sup>

$$\ln w^{\vartheta} = a^{\vartheta} + \sum_{j} \beta_{j}^{\vartheta} X_{j}^{\vartheta} + \sum_{\kappa} \beta_{\kappa}^{\vartheta} X_{\kappa}^{\vartheta} + \varepsilon^{\vartheta}$$
<sup>(2)</sup>

*j* now indicates individual characteristics and *k* employer or job characteristics.<sup>15</sup> We perform this exercise in order to compare the results of the two specifications and examine to what extent our understanding of the sources of earnings variations is refined when we control for both worker and workplace heterogeneity as opposed to controlling for worker heterogeneity only. We next continue with the analysis and the interpretation of the detailed decomposition results. These decomposition results are based on the estimation of our preferred specification, the extended specification 2 that accounts for firm heterogeneity.

#### 5.1. The added value of controlling for employer heterogeneity

Regressions using worker-based datasets typically explain about 30% of wage variation. This is also the case for the Greek results derived from Household Budget Survey data (see for instance Table 7 in Tsakloglou and Cholezas (2005)). With this as a benchmark, Table 4 presents the estimated R<sup>2</sup> values corresponding to our OLS estimations from the SES database by specification, year, and gender-group. Markedly, regressions using only individual characteristics explain 40-53% of wage variation, which is already a significant improvement in explanatory power. However, the incorporation of controls for employer/job characteristics increases the proportion explained even further, to as much as 63%. A similar pattern appears when looking at the residual standard deviation (RSD), the classic measure of within-group wage inequality a la Juhn, Murphy, and Pierce (1991, 1993), which is also reported in Table 4. Within-group or 'unexplained' wage inequality appears larger for specification 1 than for specification 2 for all years and samples.

At the same time, the inclusion of employer characteristics also alters the wage effects of the key worker characteristics. For example, if one looks at the estimated OLS coefficients (Table A1 in the Appendix), while specification 1 suggests that the return to 1 additional year of education is about 3.3% in 1995 this falls to 1.7% once

<sup>&</sup>lt;sup>15</sup> The variables used to capture individual characteristics are: years of education, tenure in years, tenure squared, age dummies, dummy for gender, dummy for vocational degree. The variables used to capture the respective employer and job characteristics are: sector dummies (2-digit NACE), occupational dummies (1-digit ISCO), size dummies, dummy for private ownership, dummies for the main market for the firms' products, regional dummies, and dummies for collective agreements.

Table 4: Estimated R <sup>2</sup> -adjusted and RSD from OLS regressions										
			1995			2002				
		All	Males	Females	All	Males	Females			
R <sup>2</sup> -adj.	Spec. 1	0.47	0.40	0.43	0.53	0.52	0.45			
	Spec. 2	0.58	0.53	0.57	0.63	0.62	0.60			
RSD	Spec. 1	0.28	0.29	0.29	0.32	0.33	0.33			
	Spec. 2	0.25	0.25	0.28	0.28	0.29	0.30			

the regression is estimated using information on both employers and employees in specification 2.

Such differences between the two specifications are also manifest in the results of the quantile regressions.<sup>16</sup> Briefly, in line with Choleza's (2004) findings for Greece and similar findings for other countries (e.g. Machado and Mata 2005 for Portugal; Izquierdo and Lacuesta 2006 for Spain), the returns to education appear to increase across the wage distribution in both specifications.<sup>17</sup> Also, they increase across time at every point of the distribution in both specifications. However, the increases suggested by specification 1 are always larger in magnitude than the increases suggested by specifications, they reflect a gender wage gap that increases in high paid jobs and decreases in time.<sup>18</sup> However, the magnitudes suggested are always higher for specification 1. The story is similar for the majority of the coefficients on employee characteristics.<sup>19</sup>

<sup>&</sup>lt;sup>16</sup> Detailed quantile regression results are presented in Tables A1-A6 in the Appendix.

<sup>&</sup>lt;sup>17</sup> It should be noted that this result is at odds with the findings of Martins and Pereira (2004) for Greece; they find that returns to education are higher at the lower quantiles. This counterintuitive result may be due to the hourly earnings variable they use and/or due to the fact that they do not account for employer characteristics. They use net hourly earnings. As the authors claim the latter measure is influenced by progressive taxation; this may provide inaccurate results for the returns to education for Greece - returns to education are eroded at higher wage quantiles.

<sup>&</sup>lt;sup>18</sup> Our findings on the gender wage gap are in line with a long-standing tradition, starting with Bergman's (1971, 1974) pioneering work in the early 1970s, and subsequently followed by a long list of literature. The reasons for it stretch from discrimination on the demand-side to female self-selection in certain occupations on the supply-side. Women may select occupations that require smaller human capital investment, as they anticipate shorter and less continuous work-lives, or occupations more compatible with the performance of household work or occupations that are traditionally dominated by women. Evidence on gender discrimination in the Greek labour market from a series of empirical studies also endorses the segregation assumption (e.g. Patrinos and Lambropoulos 1993; Kanellopoulos and Mavromaras 2002; Papapetrou 2004).

<sup>&</sup>lt;sup>19</sup> The results on employer characteristics, in specification 2, are in line with the finding of studies using similar data sets. For instance, we find that larger firms pay in general higher wages (cf. Arai, 2003 for evidence on Sweden; Magda *et. al* 2008 for evidence on various EU countries). Interestingly,

Qualitatively, the two specifications provide the same results for the employee characteristics but the magnitude of the effects differ. Specification 1 tends to systematically overestimate the returns to employee characteristics. This however, does not come as a surprise as specification 1 does not control for firm heterogeneity. As Hamermesh (2008) argues if one has a lot of information on workers and little on firms the results will show that worker characteristics matter a lot. On the other hand if one has a lot of information on firms the results will show that firm characteristics matter a lot. <sup>20</sup> Accounting for more heterogeneity leads to a better insight into the 'true' impact of employer and employee characteristics and specification 2 is a way towards this direction. <sup>21</sup>

Figure 2 presents the respective decomposition outcomes for each specification, i.e. the breakdown of observed wage changes into composition and price effects by decile. Interestingly, the contribution of price and composition effects differs significantly between specifications. Specification 1 underestimates the contribution of composition effects along the entire wage distribution. It tends also to overestimate the contribution of price effects at high deciles and to underestimate it at low deciles.<sup>22</sup> Specifically, in specification 1, the composition effects at the 9th decile are 0.12 log points higher than at the 1st decile, while the same difference for the price effects is 0.21. Likewise, the difference between the 9th and 5th deciles is 0.17 for composition effects at the 9th decile are 0.22 log points higher than at the 1st decile, while the same difference of the price effects is 0.05; the respective differences between the 9th and the 5th deciles are 0.24 for composition and 0.02 for price effects. An important common result of the two specifications should, however, be

the relative wage loss of the employees working in smaller firms is higher at higher wage deciles. Also, workers covered by a firm level agreement seem to earn higher wages (cf. Magda *et. al.* 2008). The results are not presented due to space consideration but are available upon request.

<sup>&</sup>lt;sup>20</sup> For instance, studies analysing the impact of firm profitability on wages show that the effect of firm profits on wages is significantly lower when estimated from regressions that control for both firm and employee characteristic than when estimated from regressions that control for firm characteristics only (cf. Arai, 2003; Nekby 2003).

<sup>&</sup>lt;sup>21</sup> Arguably, some of the variables capturing firm and job characteristics may be endogenous (outcome variables) and thus lead to selection bias. Among the variables included the occupation variable may perhaps be considered as endogenous (see Angrist and Pischke, 2009 for a rule of thumb for detecting 'outcome' variables). We, however, proceed with the analysis of specification 2 as, in an accounting exercise like ours that does not exclusively aim at estimating the causal wage effect of occupational status, it is constructive to know what part of the change in the wages, at different deciles, can be explained by firm, job and worker returns and characteristics (cf. Albrecht et al. 2003).

 $<sup>^{22}</sup>$  The composition and price effects from specification 2 are due to both employee and employer/job characteristics.

acknowledged. Both specifications suggest that 'price' effects have been favourable for wages throughout the wage distribution, and that the slight wage-falls that took place at the lower deciles are attributable to the negative composition effects.

Controlling for employer characteristics is thus essential not only for obtaining an insight into the 'true' returns to skills but also for identifying the factors that shape the changes in the wage distribution. The role of price/return effects would have been overestimated if inference was based on a model that did not control for the observable firm heterogeneity.



Figure 2: Breakdown of wage changes into composition and price effects by decile

# 5.2. Aggregate price and composition effects

We now proceed to the analysis of the results of specification 2. First, we briefly look into the aggregate composition and return effects of employee and employer/job characteristics. As Figure 2 indicates composition effects are negative except for the upper deciles. Therefore, they have a considerable inequality increasing effect as they tend to push wages downwards for those at the lower deciles of the wage distribution and raise them for those at high deciles. Hence, even if the returns to the employers' and employees' characteristics had remained constant, compositional changes would imply a significant increase in wage inequality. Moreover, even though the overall return effects have been positive throughout the distribution, they tend to favour slightly those at high wage deciles; they thus also have a mild inequality increasing effect.

What is also interesting is the difference in the importance of price and composition effects between men and women. When looking at men only, composition effects are larger in absolute value at most deciles; for women, though, it is the price effects that dominate. In fact, the domination of the price effects is strong enough to make no allowance for wage falls. Thus, the changes in the wage distribution among women are not driven by their changing composition but rather by increases in the returns to their labour market characteristics. These results are broadly indicative of the fact that women are catching up.

Next we continue to analyse in detail the contribution of price and composition effects of skills. Knowing how the composition and return effects of skills influence wage changes and wage inequality is important. Firstly, this can be a potential useful starting point for assessing now and in the future the impact of policies aiming at increasing the educational attainment of the labour force or of policies that increase job mobility and affect tenure and the formation of firm-specific human capital (i.e. more flexible employment contracts). Secondly, it would be also interesting to know whether the various returns to skill effects follow patterns that are in line with the skill-biased technical change or the variants of it.

#### 5.3. Wage changes due to skill

Table 5 presents in turn the breakdown of wage changes into price and composition effects of each employee characteristic/skill obtained from the full sample estimation and the separate estimation by gender. The list of employee characteristics we control for is: age as a proxy of general labour market experience,

years of education, tenure as an indicator of job-specific experience<sup>23</sup>, a dummy variable for holders of vocational degrees and a dummy variable for females (in the full sample estimations).<sup>24</sup>

The decomposition results for age are presented in two categories: the youth or minimal experience category that refers to employees with years of age between 15 and 24, and the prime-age adult or medium-high experience category that refers to employees with years of age between 25 and 54; the latter category aggregates the effects of three age-bundles (i.e., 25-34, 35-44, 45-54).<sup>25</sup> The decomposition results for tenure include the combined effects of tenure and tenure-squared.

		P10	P20	P30	P40	P50	P60	P70	P80	P90
Observed wage of	change	0214	0141	0080	0077	0074	.0480	.0997	.1474	.1953
Total composition	on effects	0819	0682	0794	0936	0921	0522	0002	.0596	.1452
of which due to:	Age (if < 25)	0096	0175	0062	0064	0068	.0012	0006	0007	0023
	Age (if > 24)	.0056	.0016	0047	0005	0065	0119	0074	0074	.0035
	Education	.0106	.0172	.0151	.0150	.0134	.0214	.0108	.0202	.0308
	Tenure	0567	0581	0580	0668	0803	0574	0495	0331	.0088
	Vocat. degree	0003	0003	0006	0006	0006	0001	0002	.0003	.0000
	All skills	0504	0571	0544	0593	0808	0468	0469	0207	.0408
	Sex (female)	.0010	.0010	0077	0086	0018	0123	0159	0138	0103
Total price effec	ts	.0349	.0418	.0512	.0647	.0683	.0829	.0948	.0847	.0888
of which due to:	Age (if < 25)	0050	0045	0039	0017	0010	0006	0008	0002	0002
	Age (if > 24)	0280	0523	0533	0603	0690	0666	0619	0600	0457
	Education	.0395	.0359	.0306	.0263	.0223	.0296	.0273	.0311	.0341
	Tenure	.0240	.0185	.0118	.0101	.0069	.0083	.0134	.0176	.0211
	Vocat. degree	0012	0016	0029	0041	0049	0070	0070	0086	0061
	All skills	.0293	0040	0177	0279	0457	0363	0290	0201	.0032
	Sex (female)	.0067	.0155	.0154	.0117	.0117	.0087	.0085	.0065	.0027

<sup>&</sup>lt;sup>23</sup> Age and tenure are separately included in the regression to account for the wage effects of general labour market experience and the effects of firm-specific experience respectively (e.g. Machado and Mata, 2005; Izquierdo and Lacuesta 2006). One could argue that there is some association between age/experience and tenure. In large cross-sections, however, this association is weakened by the fact that different individuals may have different levels of labour market experience when entering their current job.

<sup>&</sup>lt;sup>24</sup> The information on age is given in age categories/bundles rather than in years; thus regressions include a dummy variable per age bundle.

<sup>&</sup>lt;sup>25</sup> Employees between 55-64 years of age constitute the reference category.

		P10	P20	P30	P40	P50	P60	P70	P80	P90
Observed wage change		0457	0412	0297	0129	.0269	.0835	.1237	.1741	.2111
Total composition	on effects	1066	0920	0905	0666	0804	.0345	.0537	.1389	.1820
of which due to:	Age (if < 25)	0142	0215	0085	.0017	0042	0038	0014	0008	0008
	Age (if > 24)	0008	0061	0129	0038	0145	0138	0058	.0091	.0111
	Education	.0098	.0295	.0228	.0158	.0134	.0229	.0193	.0168	.0288
	Tenure	0619	0723	0739	0701	0757	0464	0357	.0016	.0233
	Vocat. degree	0005	0013	0004	0004	0002	.0000	.0002	.0000	.0017
	All skills	0677	0718	0730	0567	0812	0413	0234	.0267	.0641
Total price effec	ts	.0358	.0342	.0515	.0533	.0590	.0442	.0619	.0507	.0331
of which due to:	Age (if < 25)	0044	0021	0017	0014	0004	.0000	.0000	.0000	.0000
	Age (if > 24)	0247	0579	0499	0598	0652	0637	0637	0613	0452
	Education	.0367	.0330	.0335	.0255	.0403	.0375	.0525	.0471	.0400
	Tenure	.0191	.0143	.0106	.0008	.0009	.0027	.0076	.0177	.0329
	Vocat. degree	0034	0031	0044	0065	0079	0088	0128	0087	0111
	All skills	.0233	0158	0119	0414	0323	0323	0164	0053	.0166

Table 5b: Composition and price effects due to employee characteristics by decile, males

Table 5c: Composition and price effects due to employee characteristics by decile, females

		P10	P20	P30	P40	P50	P60	P70	P80	P90
Observed wage	change	.0006	.0206	.0367	.0431	.0605	.0739	.1089	.1627	.2135
Total composition	on effects	0657	0513	0436	0539	0423	0142	0148	.0681	.1207
of which due to:	Age (if < 25)	0035	.0000	0045	0046	0023	0049	0015	.0022	.0020
	Age (if > 24)	0007	0011	0082	0014	0032	.0029	.0013	.0007	.0146
	Education	.0073	.0194	.0160	.0125	.0145	.0156	.0115	.0215	.0127
	Tenure	0451	0556	0384	0466	0526	0436	0386	0172	.0069
	Vocat. degree	.0004	0001	0002	0007	0011	0005	0007	0001	.0000
	All skills	0416	0374	0354	0407	0447	0306	0280	.0071	.0362
Total price effec	ts	.0224	.0527	.0671	.0826	.0883	.0963	.1123	.1168	.1303
of which due to:	Age (if < 25)	0089	0078	0060	0046	0041	0034	0040	0016	0020
	Age (if > 24)	0355	0516	0642	0730	0650	0714	0929	0914	0679
	Education	.0410	.0484	.0333	.0288	.0210	.0181	.0184	.0251	.0186
	Tenure	.0262	.0270	.0225	.0232	.0283	.0280	.0243	.0174	.0222
	Vocat. degree	.0000	0002	0004	0003	0006	0006	0009	0010	0029
	All skills	.0227	.0158	0148	0260	0203	0293	0552	0515	0320

Looking, first, at the contribution of the skill composition effects, already provides confirmation of their leading role in the determination of wage changes. The

estimated composition effects of all skills together are negative for most deciles. In addition, they account for the largest part of overall composition effects, especially up to the 7th decile, for as long as the overall composition effects are negative. Their relative significance is much lower at the two highest wage deciles, where the key role is played by employer characteristics.<sup>26</sup>

The negative part of the skill composition effects appears to come primarily from the tenure variable. The negative tenure effects are, in turn, a good reflection of the decrease in the per capita levels of job-specific experience in the sample, at all but the last two wage deciles.<sup>27</sup> Tenure composition effects tend to decrease wages relatively more in the lower deciles with a consequent push towards higher wage inequality. In contrast, age effects do not have an equally noticeable contribution to the way the wage distribution evolved. The respective composition effects are smaller in magnitude and follow no regular pattern across the distribution.

Education is the only skill that has had a positive composition effect at all wage deciles. However, had returns to education remained constant, the increase in educational attainment would have lead to wage increases across the broad but it would have also led to more wage inequality. Specifically, for the aggregate sample, the composition effect of education at the 9th wage decile is 2.90 times higher than that at the 1st decile. For males, the respective ratio is 2.93 and, for females, it is 1.74. This result is rather surprising, given that rising education increases the proportion of the skilled in the labour force and induces their relative wages to fall pushing towards lower wage inequality. However, there is another factor to consider. Wages tend to be more dispersed among high skilled workers and, therefore, rising education also pushes towards more wage inequality. Which effect dominates is an empirical issue. For the case of Greece in the period 1995-2002 the evidence seems to be consistent with the latter effect being the dominant one. Our results are similar with the ones provided for Spain over the same period by Izquierdo and Lacuesta (2006) and for Portugal over 1986-1995 by Machado and Mata (2005).

<sup>&</sup>lt;sup>26</sup> The disaggregation of the overall composition effect into its components reveals that it is mainly the aggregate employer/job composition effects that contribute to the wage increases at the upper deciles of the wage distribution. For the complete accounting of the decomposition results see Table A7 in the Appendix.

<sup>&</sup>lt;sup>27</sup>The evolution of mean tenure by decile is presented in Figure A1 in the Appendix, along with all the (bootstrapped) mean employee characteristics ( $X_{it}$ ).

We turn now our focus on the price effects of skills. As already mentioned, aggregate price effects have had a significant contribution to overall wage changes, but their contribution to wage inequality has been relatively mild as they have been more equally spread across the distribution. However, if one looks at the price effects due to skill only observes a rather different picture. Evidently, even though overall price effects appear to have pushed wages upwards, the price effects due to skill, while smaller in magnitude than the skill composition effects, are also mostly negative (except for those at the 1<sup>st</sup> and 9<sup>th</sup> decile). Further, they tend to be larger in absolute value at the middle wage deciles.<sup>28</sup> For clarity of exposition, Figure 3 plots the price effects due to education, general labour market experience of prime age adults  $(age > 24)^{29}$  and job-specific experience (tenure) by decile and gender. Total skill price effects are also plotted, indicated by the shaded area. Noticeably, the total price effects of skills form a U-shaped pattern across the wage distribution.<sup>30</sup> Thus, had the composition of skills remained constant the change in the returns to skills would tend to decrease the wages of almost all employees but mostly of those in the middle of the wage distribution. It was mentioned earlier that the increase in the overall wage inequality is driven mainly by the higher upper tail inequality and that the increases in the lower tail inequality were more contained. Obviously, the Ushaped pattern of skill return effects contributes to this pattern of wage inequality.

This pattern of return effects may be associated with a deteriorating market valuation of the skills of employees in the middle of the wage distribution relative to those of employees in the lower and upper part of the distribution. This U-shaped

<sup>&</sup>lt;sup>28</sup> The disaggregation of the overall return effects into their components indicates that the constant mainly contributes to the observed wage increases. On the other hand, the return effects of employee and employer/job characteristics tend to decrease wages at most deciles. (see Table A7 in the Appendix). The fact that the effects of the constant seem to determine the sign of the overall return effects should not be considered surprising. Melly (2005) in his analysis for the US over the period 1973-1989 shows that the negative coefficient effects on the median wage change are mainly due to the negative constant effect and not due to the sign of the return effects.

<sup>&</sup>lt;sup>29</sup> In Tables 5a-5c the age return effects of both younger workers (age<25) and prime age adults (age>24) are presented and both are negative. In the Figures, though, we focus on the analysis of the price effects of prime age adults who are expected to have completed their education and be more permanently attached to the labour market (cf. Beaudry and Green, 2000). In any case, since the age return effects of younger workers are very small the thrust of the results would not change.

<sup>&</sup>lt;sup>30</sup> This U-shaped pattern is also reflected in the changes of the estimated returns from the regressions over 1995-2002 for most skill variables (see Figure A2 in the Appendix). Interestingly, our estimates of the changes in the returns to education over 1995-2002 differ from the equivalent estimates of Cholezas (2004) over 1994-1999. In contrast to our U-shaped pattern, Cholezas finds that changes in the returns to education increase monotonically when moving along the wage distribution. However, apart from the difference in the period under study, Choleza's findings are derived without controlling for employer characteristics.

pattern can be considered as being broadly in line with the routinization hypothesis. The routization hypothesis refers to technical change that tends to substitute mediumskill jobs. Therefore, the decreasing return to skills for the employees in the middle part of the distribution can be, partly, the outcome of a lower relative demand for the skills of these employees.<sup>31</sup>



Figure 3: Changes in price effects due to skill by decile

When men and women are examined separately the following interesting results emerge. For the sample of males the U-shape pattern is reinforced by a more pronounced positive part at the upper deciles of the distribution. For women the estimated price-effects display a downward trend when moving along the wage distribution, with the price effects at the low and upper wage deciles keeping their advantage in comparison to the price effects in the middle of the distribution; however the positive price effects observed for men at the upper deciles are not sustained. This could be taken as evidence of the so called 'glass-ceiling'.<sup>32</sup>

However, the evolution of the aggregate skill price-effects across wage deciles only draws part of the picture. Equally important is to examine the evolution of the

 $<sup>^{31}</sup>$  A formal test of the routinization hypothesis would require an analysis of relative employment quantities. A direct test of any hypothesis is, however, beyond the purpose of the current study.

<sup>&</sup>lt;sup>32</sup> 'Glass-ceiling' refers to the situation where the labour market performance of women follows that of men up to a point after which their wages fall behind those of men (see Albrecht et. al., 2003 and Albrecht et. al., 2009). Positive return effects of skills, in the upper deciles, that are present only in the case of men could be a manifestation of this situation.

price effects of each skill separately. Interestingly, the U-shaped pattern is more or less sustained for all individual skill price effects. In all samples and deciles, the price effects of education and tenure are positive while the respective return effects of age are negative and relatively higher in absolute value. <sup>33</sup> Thus, if the education and tenure distribution had remained constant the change in their returns would tend to increase wages. On the other hand, if the age composition of the labour force was held constant the change in the returns to age would tend to decrease wages.

Institutional changes that are effectively not age neutral is one of the explanations usually put forward in the literature for the deteriorating labour market outcomes of younger workers.<sup>34</sup> Given that the employment relationships of older workers are more or less settled the new arrangements are likely to have influenced the evolution of earnings of younger workers more (see Rosolia and Torini, 2007 who provide an analysis for Italy). As mentioned in previous sections important changes relating mainly to employment contract arrangements have taken place in Greece the period under investigation. <sup>35</sup> Therefore, changes in the institutional environment could possibly lie behind the observed negative age effects. Interestingly, the age return effects are not uniform across the wage distribution and tend to be more negative for those in the middle of the distribution. In relation to the shape of age return effects a supplementary to the routinization hypothesis explanation could be provided. In particular, the institutional changes that took place in Greece are not of the nature that would harm low wage earners, as minimum wages are still in effect and the system can be considered as being broadly regulated (see Du Caju et. al., 2008). However, the flexible employment contract arrangements in interaction with differences in unobserved labour market relevant skills may lead to divergent labour

<sup>&</sup>lt;sup>33</sup> We have also decomposed the age return effects of the prime age adults into their components and observed that the age return effects are negative for all age categories. However, for the younger workers (25-34) and those between 35-44 years of age the effects are, in most deciles, relatively more negative; interestingly, the U-shape pattern is driven to a large extent by these two age categories.

<sup>&</sup>lt;sup>34</sup> In many countries institutional changes that aim at increasing labour market flexibility entail the inception of fixed-term contracts. The wages of employees working under temporary contracts are found to be on average lower than those of employees working under permanent contracts (eg. De la Rica and Felgueroso, 2003). Further, recent evidence for Portugal shows that temporary employment contracts tend to be concentrated to younger and less educated workers (Portugal and Varejao, 2009).

<sup>&</sup>lt;sup>35</sup> Machado and Mata (2005) argue that changes in workers' unobserved ability or in institutions may be reflected in coefficient changes. Arguably, in our case the age dummies may also capture, apart from returns to general labour market experience, unobserved characteristics of the workers in these age categories or institutional changes that have an impact on these worker categories.

market outcomes for those in the middle and the upper part of the distribution.<sup>36</sup> In particular, any higher unobserved ability of those in the upper part of the distribution would tend to outweigh to some extent the impact of non-favourable institutional arrangements.

### 5.4. Economic developments versus predetermined changes

We have now seen the contribution of skills to wage-inequality in detail. But, how do these results inform us about the relative importance of economic developments and predetermined composition effects in shaping wage inequality the period under investigation?

To examine the relative importance of economic developments we need to add to the skill price effects, which are undoubtedly influenced by market conditions, the part of the skill composition effects that could also be responsive to economic developments. In the group of our skill-variables, we identify tenure as the only one whose composition effects are not strictly predetermined; in fact, we expect the market unresponsive part of tenure composition to be low.<sup>37</sup> For instance, tenure may be responsive to changes in the institutional environment that increase job mobility (i.e. inception of contracts of definite length and contracts that relate to the completion of a specific task).<sup>38</sup> Further, a non negligible part of tenure composition effects is determined by the employers according to market conditions. In periods of high demand, we expect reduced firing to increase tenure and increased hiring to decrease it, and vice versa in periods of low demand.

<sup>&</sup>lt;sup>36</sup> Differences in unobserved labour quality have been frequently proposed as an explanation for wage differences between groups of workers. Bound and Johnson (1992) argue that unobserved labour quality in interaction with technical change can explain the change in the structure of wages in the US in the 1980s. In a similar reasoning, Albrecht *et. al.* (2009) argue that the low dispersion in labour market relevant skills in Sweden may lie behind the contained increases in wage inequality.

<sup>&</sup>lt;sup>37</sup> One could of course argue that education and age composition may also be responsive to economic developments and market conditions. For instance, increasing returns to education could affect incentives and lead employees to acquire more education (e.g Acemoglou 2002). Further, an increase in the number of years devoted to education will in turn postpone the entry to the labour market; therefore, the age composition of the labour force may also be influenced. However, this is a process that is not instant and takes time. Given, the short time period covered by our sample it is not clear what part of the changes in education and age is exogenous and what part is responsive to economic developments. We prefer to treat age and education as exogenous and consider the responsive to economic developments skill effects obtained as a lower bound of the respective effects.

<sup>&</sup>lt;sup>38</sup> We have referred in the previous section to the possible impact of institutional changes on the pattern of age price effects. As mentioned here tenure levels may also be influenced by some of these institutional changes. In such case, the age price effects are likely to capture the impact of these institutional arrangements on wages that is beyond the one relating to tenure.

Hence, we aggregate the total price-effects of skill with the tenure composition effect and call it the responsive to market conditions and economic developments part of skill effects. Then, in Figure 4, we plot it against the sum of the skill composition effects that are exclusively attributable to predetermined demographic changes – education and age composition effects. One can see that the responsive to economic developments skill effects are important in magnitude; in contrast, the predetermined skill composition effects are of relatively smaller scale. Interestingly, the U-shape pattern of return effects to the skill price-effects; it should be pointed though that the slightly positive skill price effects in the lower and upper deciles are now outweighed and a positive part is only present for males and only at the upper deciles.

Figure 4 also plots the overall observed wage changes (indicated by the shaded area). This serves to illustrate another important point. The skill effects attributable to economic developments have also contributed to the observed pattern of wage inequality, i.e. towards the upper-tail wage inequality and the wage-compression observed in the lower half of the distribution. Obviously, they tend to decrease more the wages of those in the middle of the distribution as compared to the wages of those in the second pattern.

Another interesting point that emerges from Figure 4 is that skills, despite playing an important role in shaping the changes in the wage distribution over the period 1995-2002, give only part of the picture. We need to also look into the effects that tend to outweigh the negative skill price and composition effects and contribute to the observed wage increases. It is the effect of the constant across the wage distribution and of the employer/job composition effects in the upper deciles that outweigh the negative skill price and composition effects making no allowance for extensive wage falls. Interestingly, the employer/job composition effects contribute also to the increases in the overall wage inequality. In particular, had the returns to employer and job characteristics remained constant their changing composition would tend to increase overall wage inequality, as they tend to decrease wages at low deciles and increase them at high deciles (see Table A7 in the Appendix). The fact that the employer/job characteristics are also very likely responsive to market conditions further reinforces the case for the importance of economic developments in influencing the pattern of wage changes. Conclusively, we are more inclined to side

against the 'revisionists' view, in general, and the conclusion of Tsakloglou and Cholezas (2005) about Greece in particular, which assign economic developments a secondary role in the determination of wage inequality.<sup>39</sup>



Figure 4: Skill-effects by source and observed wage changes

# 6. Conclusion

In this paper we have examined how the distribution of wages has changed in Greece over 1995-2002 and what has been the contribution of skills to these changes. We have used a matched employer-employee dataset, which allowed us to control for both worker and job/employer heterogeneity. Building on the regression results, we used the Machado-Mata decomposition method to separate the part of the wage changes that is due to changes in the employer/job and workers characteristics from the part due to changes in the returns to these characteristics. Ultimately, this enabled us to analyse the contribution of the price and composition effects of each skill separately and join together the skill price effects and the part of the composition

<sup>&</sup>lt;sup>39</sup> Christopoulou, Jimeno and Lamo (2010) in their analysis of nine EU countries covering the same period and using the Structure of Earning Survey also find that economic developments are the most relevant in explaining the observed wage dynamics.

effects of skills that is responsive to market conditions. The latter exercise proves useful in our attempt to examine the relative importance of economic developments and predetermined compositional changes in shaping the observed changes in the wage distribution.

The evidence suggests a small increase in Greek average wages combined with a significant increase in wage inequality, mostly due to significantly higher relative wage increases at the upper tail of the distribution. The contribution of skills to the evolution of Greek wage dynamics has been important. Briefly, the composition effects of skills are mostly negative. Their negative part is mainly due to the tenure composition effects which, by reducing wages more for those in the lower part of the wage distribution, contribute to the observed pattern of wage inequality. Markedly, the composition effects of education despite being positive also contribute to higher wage inequality. This result contradicts standard expectations for a negative relationship between rising education and wage equality and it is possibly the outcome of higher wage dispersion among the high skilled.

The price effects due to skill have also been mostly negative but their magnitude is relatively smaller. Interestingly, they have formed a U-shaped pattern along the wage distribution. This U-shaped pattern of skill price effects, though not a proof of, can be considered as being broadly in line with the routinization hypothesis; a variant of the of skill-biased technical change theory. As to the individual skill price effects, tenure and education price effects are positive and the negative part of the skill price effects is mainly due to the age price effects. The sign and shape of the age price effects could be regarded as being the outcome of the interaction between institutional changes that are not age neutral and different unobserved ability of employees along the wage distribution.

Noticeably, it is the effects of the constant across the wage distribution and of the employer/job composition effects in the upper deciles of the wage distribution that outweigh the negative skill price and composition effects making no allowance for extensive wage falls. In interpreting our decomposition results we can argue that economic developments, through their combined influence on the price effects of skills and the market driven part of the skill composition effects have had an important role in shaping Greek wage dynamics. On the other hand, the skill effects that are attributable to predetermined compositional changes were of relatively smaller scale.
## References

Abowd, J. M. and F. Kramarz (1999), "The Analysis of Labor Markets Using Matched Employer-Employee Data" in Ashenfelter, O. and D. Card (eds.), Handbook of Labor Economics, Vol. 3B, p. 2629-2710, North Holland, Amsterdam.

Acemoglu, D. (2002), "Technical Change, Inequality, and the Labor Market", *Journal of Economic Literature*, XL, p.7-72.

Albrecht, J., A. Bjorklund and S. Vroman (2003), "Is there a Glass Ceiling in Sweden?", *Journal of Labour Economics*, 21, p. 145-177.

Albrecht, J., A. Bjorklund and S. Vroman (2009), "Unionization and the Evolution of the Wage Distribution in Sweden: 1968 to 2000", IZA Discussion Paper No. 4246.

Angrist J. D. and J. F. Pischke (2009), *Mostly Harmless Econometrics*, Princeton University Press, Princeton.

Arai, M. (2003), "Wages, Profits and Capital Intensity: Evidence from Matched Worker-Firm Data", *Journal of Labor Economics*, 21, p 593-618.

Autor, D. H., L. F. Katz and M. S. Kearney (2008), "Trends in Wage Inequality: Revisiting the Revisionists", *Review of Economics and Statistics*, 90, p. 300-323.

Autor, D. H., L. F. Katz and M. S. Kearney (2006), "The Polarization of the U.S. Labor Market", *American Economic Review*, 96, p. 189-194.

Autor, D. H., L. F. Katz and M. S. Kearney (2005), "Rising Wage Inequality: The Role of Composition and Prices" NBER Working Paper No. 11628.

Autor, D. H., F. Levy and R. J. Murnane (2003), "The Skill Content of Recent Technological Change: An Empirical Exploration", *Quarterly Journal of Economics*, CXVIII, p. 1279-1333.

Beaudry, P. and D. A. Green (2000), "Cohort Patterns in Canadian Earnings: Assessing the Role of Skill Premia in Inequality Trends", *Canadian Journal of Economics*, 33, p. 907-936.

Bergmann, B. R. (1971), "The Effect on White Incomes of Discrimination in Employment", *Journal of Political Economy*, 79, p. 294-313.

Bergmann, B. R. (1974), "Occupational Segregation, Wages and Profits when Employers Discriminate by Race or Sex", *Eastern Economic Journal*, 1, p. 103-110.

Blanchard, O. and J. Wolfers (2000), "The Role of Shocks and Institutions in the Rise of European Unemployment: the Aggregate Evidence", *Economic Journal*, 110, p.C1-C33.

Blinder, A.S. (1973), "Wage Discrimination: Reduced Form and Structural Estimates" *Journal of Human Resources*, 8, p. 436-455.

Bound, J. and G. Johnson (1992), "Changes in the Structure of Wages in the 1980's: An Evaluation of Alternative Explanations," *American Economic Review*, 82, p. 371–392

Card, D. and J. E. DiNardo (2002), "Skill-Biased Technological Change and Rising Wage Inequality: Some Problems and Puzzles", *Journal of Labor Economics*, 20, p. 733-783.

Cholezas, I. (2004), "Private Returns to Education in Greece: A Quantile Regression Analysis." Department of International and European Economic Studies, Athens University of Economics and Business, mimeo.

Christopoulou, R., J. F. Jimeno and A. Lamo (2010), "Changes in the Wage Structure in EU countries", ECB Working Paper No.1199.

Conway, P., V. Janod and G. Nicoletti (2005), "Product Market Regulation in OECD Countries: 1998 to 2003", OECD Economics Department Working Paper No. 419.

De la Rica, S. and F. Felguerso (2003), "Wage Differentials Between Permanent and Temporal Workers: Further Evidence", DFAEII Working Paper No. 200207.

DiNardo, J., N. Fortin and T. Lemieux (1996), "Labour Market Institutions and the Distribution of Wages, 1973-1992: A Semiparametric Approach", *Econometrica*, 64, p. 1001-1044.

Du Caju, P, E. Gautier, D. Momferatou and M. Ward-Warmedinger (2008), "Institutional Features of Wage Bargaining in EU countries, the US and Japan", ECB Working Paper No. 974.

Dustmann, C. and C. Meghir (2005), "Wages, Experience and Seniority" *Review of Economic Studies*, 75, p. 77-108.

Fiori G, G. Nicoletti, F. Schiantarelli and S. Scarpetta (2008), "Employment Outcomes and the Interaction Between Product and Labor Market Deregulation: Are They Substitutes or Complements?", paper presented at the 5th ECB/CEPR Labour Market Workshop.

Goos, M. and A. Manning (2007), "Lousy and Lovely Jobs: the Rising Polarization of Work in Britain", *Review of Economics and Statistics*, 89, p. 118-133.

Goos, M. A. Manning and A. Salomons (2009), "Job Polarization in Europe", *American Economic Review*, 99, p. 58-63.

Hamermesh, D. (2008), "Fun with Firm-Employee Data: Progress and Road Maps", *Labour Economics*, 15, p. 663-673.

Izquierdo, M. and A. Lacuesta (2006), "Wage Inequality in Spain: Recent Developments", Bank of Spain Working Paper No. 0615.

Juhn, C., K. Murphy, and B. Pierce (1991), "Accounting for the Slowdown in Black-White Wage Convergence" in M. Kosters (ed.), Workers and their Wages, p. 107-143, American Enterprise Institute, Washington DC.

Juhn, C., Murphy, K. M. and B. Pierce (1993), "Wage Inequality and the Rise in Returns to Skill", *Journal of Political Economy*, 101, p. 410-442.

Kanellopoulos, C. N. and K. G. Mavromaras (2002), "Male-Female Labour Market Participation and Wage Differentials in Greece", *Labour*, 16, p.771-801.

Katz, L. F. and D. H. Autor (1999), "Changes in the Wage Structure and Earnings Inequality" in Ashenfelter, O. and D. Card (eds.), Handbook of Labor Economics, Vol 3A, p. 1464-1555, North Holland, Amsterdam.

Krugman, P. (1994), "Past and Prospective Causes of High Unemployment", *Federal Reserve Bank of Kansas City Economic Review*, 79, p.23-43.

Lemieux, T. (2006), "Increasing Residual Wage Inequality: Composition Effects, Noisy Data, or Rising Demand for Skill?", *American Economic Review*, 96, p. 461-498.

Machado, J.A.F. and J. Mata (2005), "Counterfactual Decomposition of Changes in Wage Distributions Using Quantile Regression", *Journal of Applied Econometrics*, 20, p. 445-465.

Machin S. and J. Van Reenen (2007), "Changes in Wage Inequality", CEP Special Paper No.18.

Magda, I., F. Rycx, I. Tojerow and D. Valsamis (2008), "Wage Differentials Across Sectors in Europe: An East-West Comparison", IZA Discussion Paper No. 3830.

Martins, P. and T. P. Pereira (2004) "Does education reduce wage inequality? Quantile Regression Evidence from 16 Countries", *Labour Economics*, 11, p. 355-371.

Melly, B. (2006), "Estimation of Counterfactual Distributions Using Quantile Regression" Swiss Institute for International Economics and Applied Economic Research, mimeo.

Melly, B. (2005), "Decomposition of Differences in Distributions Using Quantile Regression", *Labour Economics*, 12, p. 577-590.

Nekby, L. (2003), "Gender Differences in Rent Sharing and its Implications for the Gender Wage Gap, Evidence for Sweden" *Economic Letters*, 81, p. 403-410.

Oaxaca R. (1973), "Male-Female Differentials in Urban Labor Larkets", *International Economic Review*, 14, p. 693-709.

Papapetrou, E. M. (2004), "Gender Wage Differentials in Greece", Bank of Greece Economic Bulletin, 23, p. 47-64.

Patrinos, H. A and H. Lambropoulos (1993), "Gender Discrimination in the Greek Labour Market", *Education Economics*, 1, p. 153-164.

Portugal, P. and J. Varejao (2009), "Why do Firms Use Fixed-Term Contracts?", IZA Discussion Paper No. 4380.

Rosolia, A. and R. Torrini (2007), "The Generation Gap: Relative Earnings of Young and Old Workers in Italy", Bank of Italy Working Paper No. 639.

Sabethai, I. D. (2000), "The Greek Labour Market: Features, Problems and Policies", Bank of Greece Economic Bulletin, 16, p. 7-40.

Tsakloglou, P. and I. Cholezas, (2005), "Education and Inequality in Greece", IZA Discussion Paper No. 1582.

Zografakis, S., A. Kontis and T. Mitrakos (2009), "The Steps of Immigrants in the Greek Economy", Institute for Migration Policy, Athens (in Greek).



## A Appendix

	OLS	1	2	3	4	stimations	6	7	8	9
		1	2	5			0	,	0	,
						fication 1				
Ain. years of education		0.0216	0.0242	0.0266	0.0285	0.0311	0.0336	0.0362	0.0390	0.0426
	[0.0004]***	[0.0004]***	[0.0004]***	[0.0004]***	[0.0005]***	[0.0005]***	[0.0006]***	[0.0006]***	[0.0007]***	[0.0009]**
ocational degree	0.0678	0.0879	0.0879	0.0799	0.0831	0.0804	0.0761	0.0652	0.0498	0.0334
	[0.0060]***	[0.0068]***	[0.0065]***	[0.0060]***	[0.0071]***	[0.0075]***	[0.0076]***	[0.0084]***	[0.0095]***	[0.0114]**
Age:15-24 years old	-0.2806	-0.1407	-0.1608	-0.1907	-0.2239	-0.2530	-0.2792	-0.3050	-0.3381	-0.4094
	[0.0086]***	[0.0096]***	[0.0092]***	[0.0086]***	[0.0101]***	[0.0107]***	[0.0109]***	[0.0121]***	[0.0138]***	[0.0168]*
ge:25-34 years old	-0.1258	-0.0636	-0.0701	-0.0789	-0.0975	-0.1104	-0.1216	-0.1388	-0.1433	-0.1693
	[0.0064]***	[0.0072]***	[0.0070]***	[0.0064]***	[0.0076]***	[0.0081]***	[0.0082]***	[0.0092]***	[0.0105]***	[0.0127]*
ge:35-44 years old	0.0051	0.0378	0.0381	0.0397	0.0223	0.0154	0.0081	-0.0019	-0.0013	-0.0181
	[0.0061]	[0.0068]***	[0.0065]***	[0.0060]***	[0.0072]***	[0.0076]**	[0.0077]	[0.0085]	[0.0097]	[0.0117]
ge:45-54 years old	0.0475	0.0519	0.0605	0.0613	0.0535	0.0509	0.0476	0.0422	0.0426	0.0402
	[0.0060]***	[0.0068]***	[0.0065]***	[0.0060]***	[0.0070]***	[0.0075]***	[0.0076]***	[0.0084]***	[0.0095]***	[0.0113]*
enure in years	0.0267	0.0283	0.0286	0.0288	0.0290	0.0291	0.0283	0.0283	0.0259	0.0213
	[0.0007]***	[0.0008]***	[0.0007]***	[0.0007]***	[0.0008]***	[0.0008]***	[0.0008]***	[0.0009]***	[0.0010]***	[0.0013]*
enure squared/100	-0.0283	-0.0307	-0.0284	-0.0286	-0.0290	-0.0297	-0.0276	-0.0310	-0.0265	-0.0204
	[0.0023]***	[0.0028]***	[0.0026]***	[0.0024]***	[0.0028]***	[0.0030]***	[0.0030]***	[0.0033]***	[0.0037]***	[0.0044]*
ex: female	-0.1799	-0.1002	-0.1248	-0.1446	-0.1594	-0.1773	-0.1916	-0.2122	-0.2404	-0.2674
	[0.0032]***	[0.0037]***	[0.0035]***	[0.0032]***	[0.0037]***	[0.0040]***	[0.0040]***	[0.0044]***	[0.0049]***	[0.0059]*
onstant	1.4314	1.1521	1.2213	1.2764	1.3395	1.3936	1.4550	1.5290	1.6302	1.8066
	[0.0074]***	[0.0077]***	[0.0076]***	[0.0072]***	[0.0086]***	[0.0092]***	[0.0094]***	[0.0104]***	[0.0118]***	[0.0140]*
Observations	38071	38071	38071	38071	38071	38071	38071	38071	38071	38071
R-squared	0.47									
						fication 2				
In. years of education		0.0122	0.0137	0.0150	0.0157	0.0164	0.0164	0.0177	0.0184	0.0194
	[0.0006]***	[0.0007]***	[0.0007]***	[0.0006]***	[0.0006]***	[0.0006]***	[0.0007]***	[0.0007]***	[0.0008]***	[0.0012]*
ocational degree	0.0643	0.0549	0.0551	0.0645	0.0686	0.0692	0.0705	0.0745	0.0607	0.0727
	[0.0057]***	[0.0076]***	[0.0072]***	[0.0061]***	[0.0060]***	[0.0064]***	[0.0069]***	[0.0072]***	[0.0082]***	[0.0120]*
ge:15-24 years old	-0.2202	-0.1249	-0.1540	-0.1679	-0.1796	-0.1874	-0.2082	-0.2284	-0.2596	-0.2953
	[0.0078]***	[0.0103]***	[0.0098]***	[0.0083]***	[0.0082]***	[0.0087]***	[0.0094]***	[0.0098]***	[0.0113]***	[0.0166]*
ge:25-34 years old	-0.1031	-0.0604	-0.0692	-0.0751	-0.0824	-0.0829	-0.0947	-0.1088	-0.1191	-0.1363
	[0.0059]***	[0.0076]***	[0.0074]***	[0.0062]***	[0.0062]***	[0.0065]***	[0.0070]***	[0.0075]***	[0.0086]***	[0.0126]*
age:35-44 years old	0.0100	0.0345	0.0279	0.0236	0.0174	0.0211	0.0141	0.0067	-0.0022	-0.0107
	[0.0055]*	[0.0072]***	[0.0069]***	[0.0059]***	[0.0058]***	[0.0061]***	[0.0066]**	[0.0069]	[0.0079]	[0.0116]
Age:45-54 years old	0.0420	0.0533	0.0492	0.0447	0.0423	0.0472	0.0389	0.0363	0.0295	0.0259
	[0.0054]***	[0.0072]***	[0.0069]***	[0.0058]***	[0.0057]***	[0.0060]***	[0.0065]***	[0.0068]***	[0.0078]***	[0.0113]*
enure in years	0.0221	0.0228	0.0234	0.0237	0.0233	0.0226	0.0221	0.0208	0.0194	0.0171
	[0.0006]***	[0.0008]***	[0.0008]***	[0.0007]***	[0.0006]***	[0.0007]***	[0.0007]***	[0.0008]***	[0.0009]***	[0.0013]*
enure squared/100	-0.0197	-0.0209	-0.0205	-0.0203	-0.0183	-0.0159	-0.0155	-0.0144	-0.0148	-0.0137
* · · · · ·	[0.0021]***	[0.0030]***	[0.0028]***	[0.0023]***	[0.0023]***	[0.0024]***	[0.0026]***	[0.0027]***	[0.0030]***	[0.0045]*
ex: female	-0.1513	-0.0876	-0.1100	-0.1239	-0.1354	-0.1487	-0.1594	-0.1725	-0.1889	-0.2059
	[0.0031]***	[0.0044]***	[0.0041]***	[0.0034]***	[0.0033]***	[0.0034]***	[0.0037]***	[0.0038]***	[0.0043]***	[0.0062]*
onstant	1.6430	1.3550	1.4441	1.4700	1.4914	1.6031	1.6843	1.7810	1.8638	2.0734
onstan	[0.0346]***	[0.0436]***	[0.0433]***	[0.0369]***	[0.0360]***	[0.0381]***	[0.0407]***	[0.0432]***	[0.0491]***	[0.0690]*
bservations	37901	37901	37901	37901	37901	37901	37901	37901	37901	[0.0090] <sup>+</sup> 37901
	5/901									1/201

Table A1: OLS and quantile estimation results, Sample: all, Year: 1995

Notes: \* significant at 10%; \*\*\* significant at 5%; \*\*\* significant at 1%. Standard errors in brackets. Specification 1 controls only for the individual characteristics listed. Specification 2 also controls for observable employer and job characteristics, namely: sector, occupation, firm ownership, region, firm size, main product market and level of collective agreement coverage.

ECB Working Paper Series No 1309 March 2011

					opeer	neution 1				
Min. years of education	on 0.0498	0.0335	0.0377	0.0416	0.0450	0.0473	0.0500	0.0525	0.0551	0.0602
	[0.0005]***	[0.0006]***	[0.0005]***	[0.0005]***	[0.0006]***	[0.0005]***	[0.0007]***	[0.0007]***	[0.0009]***	[0.0011]***
Vocational degree	-0.0462	-0.0045	-0.0296	-0.0428	-0.0512	-0.0457	-0.0504	-0.0527	-0.0598	-0.0614
	[0.0057]***	[0.0077]	[0.0061]***	[0.0062]***	[0.0062]***	[0.0060]***	[0.0075]***	[0.0076]***	[0.0086]***	[0.0104]***
Age:15-24 years old	-0.3486	-0.1722	-0.1989	-0.2298	-0.2813	-0.3321	-0.3796	-0.4140	-0.4776	-0.5348
	[0.0094]***	[0.0126]***	[0.0100]***	[0.0102]***	[0.0103]***	[0.0098]***	[0.0124]***	[0.0127]***	[0.0144]***	[0.0173]***
Age:25-34 years old	-0.2141	-0.1011	-0.1211	-0.1429	-0.1774	-0.2068	-0.2404	-0.2540	-0.2851	-0.3153
	[0.0078]***	[0.0103]***	[0.0082]***	[0.0084]***	[0.0085]***	[0.0081]***	[0.0103]***	[0.0105]***	[0.0119]***	[0.0144]***
Age:35-44 years old	-0.0699	-0.0238	-0.0228	-0.0231	-0.0376	-0.0581	-0.0779	-0.0833	-0.1132	-0.1313
	[0.0075]***	[0.0100]**	[0.0080]***	[0.0081]***	[0.0082]***	[0.0079]***	[0.0010]***	[0.0101]***	[0.0114]***	[0.0138]***
Age:45-54 years old	-0.0111	0.0354	0.0372	0.0323	0.0132	-0.0081	-0.0249	-0.0321	-0.0534	-0.0601
	[0.0073]	[0.0099]***	[0.0078]***	[0.0080]***	[0.0080]*	[0.0078]	[0.0097]**	[0.0098]***	[0.0110]***	[0.0133]***
Tenure in years	0.0299	0.0315	0.0300	0.0300	0.0301	0.0308	0.0307	0.0303	0.0295	0.0269
	[0.0006]***	[0.0009]***	[0.0007]***	[0.0007]***	[0.0007]***	[0.0007]***	[0.0008]***	[0.0008]***	[0.0009]***	[0.0011]***
Tenure squared/100	-0.0216	-0.0239	-0.0107	-0.0079	-0.0085	-0.0137	-0.0188	-0.0230	-0.0282	-0.0314
	[0.0023]***	[0.0034]***	[0.0026]***	[0.0026]***	[0.0026]***	[0.0024]***	[0.0030]***	[0.0031]***	[0.0035]***	[0.0042]***
Sex: female	-0.1559	-0.0973	-0.1131	-0.1262	-0.1395	-0.1479	-0.1585	-0.1704	-0.1896	-0.2201
	[0.0034]***	[0.0047]***	[0.0037]***	[0.0037]***	[0.0037]***	[0.0035]***	[0.0044]***	[0.0045]***	[0.0050]***	[0.0061]***
Constant	1.3589	1.0599	1.1351	1.1903	1.2575	1.3330	1.4161	1.5016	1.6273	1.7845
	[0.0088]***	[0.0111]***	[0.0089]***	[0.0092]***	[0.0094]***	[0.0092]***	[0.0117]***	[0.0119]***	[0.0135]***	[0.0161]***
Observations	41449	41449	41449	41449	41449	41449	41449	41449	41449	41449
R-squared	0.53									
					*	fication 2				
Min. years of education		0.0162	0.0174	0.0180	0.0184	0.0186	0.0192	0.0201	0.0212	0.0223
	[0.0007]***	[0.0008]***	[0.0007]***	[0.0007]***	[0.0007]***	[0.0006]***	[0.0008]***	[0.0008]***	[0.0009]***	[0.0014]***
Vocational degree	-0.0101	-0.0053	-0.0080	-0.0115	-0.0093	-0.0102	-0.0122	-0.0129	-0.0216	-0.0194
	[0.0052]*	[0.0063]	[0.0057]	[0.0052]**	[0.0055]*	[0.0049]**	[0.0060]**	[0.0060]**	[0.0071]***	[0.0108]*
Age:15-24 years old	-0.3000	-0.1552	-0.2030	-0.2230	-0.2443	-0.2608	-0.2894	-0.3055	-0.3453	-0.3907
	[0.0084]***	[0.0101]***	[0.0091]***	[0.0083]***	[0.0089]***	[0.0079]***	[0.0099]***	[0.0099]***	[0.0115]***	[0.0179]***
Age:25-34 years old	-0.1864	-0.0900	-0.1305	-0.1423	-0.1558	-0.1659	-0.1798	-0.1881	-0.2073	-0.2141
	[0.0069]***	[0.0082]***	[0.0075]***	[0.0068]***	[0.0073]***	[0.0065]***	[0.0081]***	[0.0082]***	[0.0095]***	[0.0147]***
Age:35-44 years old	-0.0629	-0.0143	-0.0406	-0.0424	-0.0519	-0.0525	-0.0591	-0.0630	-0.0685	-0.0625
	[0.0067]***	[0.0080]*	[0.0072]***	[0.0066]***	[0.0071]***	[0.0063]***	[0.0078]***	[0.0078]***	[0.0091]***	[0.0140]***
Age:45-54 years old	-0.0163	0.0201	-0.0010	-0.0006	-0.0065	-0.0107	-0.0193	-0.0242	-0.0260	-0.0203
	[0.0065]**	[0.0077]***	[0.0070]	[0.0064]	[0.0068]	[0.0061]*	[0.0076]**	[0.0076]***	[0.0089]***	[0.0136]
Tenure in years	0.0228	0.0295	0.0270	0.0250	0.0240	0.0227	0.0224	0.0213	0.0199	0.0174

Table A2: OLS and quantile estimation results, Sample: all, Year: 2002

4

OLS

1

2

3

Quantile estimations

5

Specification 1

6

7

8

9

Tenure in years	0.0299	0.0515	0.0500	0.0500	0.0501	0.0508	0.0307	0.0505	0.0295	0.0209
	[0.0006]***	[0.0009]***	[0.0007]***	[0.0007]***	[0.0007]***	[0.0007]***	[0.0008]***	[0.0008]***	[0.0009]***	[0.0011]***
Tenure squared/100	-0.0216	-0.0239	-0.0107	-0.0079	-0.0085	-0.0137	-0.0188	-0.0230	-0.0282	-0.0314
	[0.0023]***	[0.0034]***	[0.0026]***	[0.0026]***	[0.0026]***	[0.0024]***	[0.0030]***	[0.0031]***	[0.0035]***	[0.0042]***
Sex: female	-0.1559	-0.0973	-0.1131	-0.1262	-0.1395	-0.1479	-0.1585	-0.1704	-0.1896	-0.2201
	[0.0034]***	[0.0047]***	[0.0037]***	[0.0037]***	[0.0037]***	[0.0035]***	[0.0044]***	[0.0045]***	[0.0050]***	[0.0061]***
Constant	1.3589	1.0599	1.1351	1.1903	1.2575	1.3330	1.4161	1.5016	1.6273	1.7845
	[0.0088]***	[0.0111]***	[0.0089]***	[0.0092]***	[0.0094]***	[0.0092]***	[0.0117]***	[0.0119]***	[0.0135]***	[0.0161]***
Observations	41449	41449	41449	41449	41449	41449	41449	41449	41449	41449
R-squared	0.53									
					Specif	fication 2				
Min. years of education	0.0204	0.0162	0.0174	0.0180	0.0184	0.0186	0.0192	0.0201	0.0212	0.0223
	[0.0007]***	[0.0008]***	[0.0007]***	[0.0007]***	[0.0007]***	[0.0006]***	[0.0008]***	[0.0008]***	[0.0009]***	[0.0014]***
Vocational degree	-0.0101	-0.0053	-0.0080	-0.0115	-0.0093	-0.0102	-0.0122	-0.0129	-0.0216	-0.0194
	[0.0052]*	[0.0063]	[0.0057]	[0.0052]**	[0.0055]*	[0.0049]**	[0.0060]**	[0.0060]**	[0.0071]***	[0.0108]*
Age:15-24 years old	-0.3000	-0.1552	-0.2030	-0.2230	-0.2443	-0.2608	-0.2894	-0.3055	-0.3453	-0.3907
	[0.0084]***	[0.0101]***	[0.0091]***	[0.0083]***	[0.0089]***	[0.0079]***	[0.0099]***	[0.0099]***	[0.0115]***	[0.0179]***
Age:25-34 years old	-0.1864	-0.0900	-0.1305	-0.1423	-0.1558	-0.1659	-0.1798	-0.1881	-0.2073	-0.2141
	[0.0069]***	[0.0082]***	[0.0075]***	[0.0068]***	[0.0073]***	[0.0065]***	[0.0081]***	[0.0082]***	[0.0095]***	[0.0147]***
Age:35-44 years old	-0.0629	-0.0143	-0.0406	-0.0424	-0.0519	-0.0525	-0.0591	-0.0630	-0.0685	-0.0625
	[0.0067]***	[0.0080]*	[0.0072]***	[0.0066]***	[0.0071]***	[0.0063]***	[0.0078]***	[0.0078]***	[0.0091]***	[0.0140]***
Age:45-54 years old	-0.0163	0.0201	-0.0010	-0.0006	-0.0065	-0.0107	-0.0193	-0.0242	-0.0260	-0.0203
	[0.0065]**	[0.0077]***	[0.0070]	[0.0064]	[0.0068]	[0.0061]*	[0.0076]**	[0.0076]***	[0.0089]***	[0.0136]
Tenure in years	0.0228	0.0295	0.0270	0.0250	0.0240	0.0227	0.0224	0.0213	0.0199	0.0174
	[0.0006]***	[0.0007]***	[0.0006]***	[0.0006]***	[0.0006]***	[0.0005]***	[0.0007]***	[0.0007]***	[0.00086]***	[0.0012]***
Tenure squared/100	-0.0184	-0.0334	-0.0239	-0.0167	-0.0149	-0.0119	-0.0124	-0.0116	-0.0113	-0.0087
	[0.0021]***	[0.0025]***	[0.0023]***	[0.0021]***	[0.0022]***	[0.0019]***	[0.0024]***	[0.0024]***	[0.0028]***	[0.0044]**
Sex: female	-0.1242	-0.0745	-0.0803	-0.0881	-0.1020	-0.1107	-0.1229	-0.1345	-0.1536	-0.1831
	[0.0032]***	[0.0040]***	[0.0036]***	[0.0032]***	[0.0034]***	[0.0030]***	[0.0037]***	[0.0037]***	[0.0041]***	[0.0066]***
Constant	1.7875	1.3543	1.4882	1.5625	1.6458	1.7302	1.8099	1.8759	2.1417	2.2502
	[0.0332]***	[0.0395]***	[0.0358]***	[0.0326]***	[0.0348]***	[0.0309]***	[0.0386]***	[0.0386]***	[0.0448]***	[0.0692]***
Observations	41449	41449	41449	41449	41449	41449	41449	41449	41449	41449
R-squared	0.63									

	OLS				Quantile esti					
		1	2	3	4	5	6	7	8	9
					Specifi	ication 1				
Min. years of education		0.0230	0.0244	0.0257	0.0278	0.0296	0.0316	0.0340	0.0352	0.0382
	[0.0005]***	[0.0006]***	[0.0006]***	[0.0006]***	[0.0006]***	[0.0006]***	[0.0007]***	[0.00086]***	[0.0010]***	[0.0012]*
Vocational degree	0.0652	0.0825	0.0848	0.0766	0.0804	0.0810	0.0720	0.0558	0.0421	0.0298
	[0.0066]***	[0.0080]***	[0.0085]***	[0.0078]***	[0.0077]***	[0.0078]***	[0.0087]***	[0.0100]***	[0.0114]***	[0.0133]*
Age:15-24 years old	-0.3391	-0.1738	-0.2154	-0.2436	-0.2806	-0.3150	-0.3567	-0.4006	-0.4440	-0.5306
	[0.0117]***	[0.0142]***	[0.0150]***	[0.0138]***	[0.0137]***	[0.0140]***	[0.0155]***	[0.0179]***	[0.0207]***	[0.0242]*
Age:25-34 years old	-0.1529	-0.0807	-0.0929	-0.1119	-0.1311	-0.1506	-0.1646	-0.1844	-0.1921	-0.2194
	[0.0078]***	[0.0093]***	[0.0099]***	[0.0091]***	[0.0091]***	[0.0093]***	[0.0103]***	[0.0120]***	[0.0139]***	[0.0163]*
Age:35-44 years old	-0.0056	0.0351	0.0321	0.0257	0.0044	-0.0066	-0.0147	-0.0256	-0.0247	-0.0331
	[0.0072]	[0.0088]***	[0.0093]***	[0.0085]***	[0.0084]	[0.0086]	[0.0095]	[0.0109]**	[0.0127]*	[0.0148]*
Age:45-54 years old	0.0531	0.0599	0.0669	0.0672	0.0547	0.0511	0.0507	0.0468	0.0494	0.0459
	[0.0070]***	[0.0086]***	[0.0091]***	[0.0083]***	[0.0082]***	[0.0084]***	[0.0092]***	[0.0106]***	[0.0122]***	[0.0142]*
Tenure in years	0.0265	0.0317	0.0311	0.0302	0.0307	0.0299	0.0282	0.0267	0.0232	0.0173
	[0.0008]***	[0.0011]***	[0.0011]***	[0.0010]***	[0.0010]***	[0.0010]***	[0.0011]***	[0.0013]***	[0.0015]***	[0.0017]*
Tenure squared/100	-0.0307	-0.0417	-0.0374	-0.0357	-0.0379	-0.0369	-0.0332	-0.0315	-0.0249	-0.0147
•	[0.0029]***	[0.0038]***	[0.0039]***	[0.0035]***	[0.0034]***	[0.0035]***	[0.0038]***	[0.0044]***	[0.0051]***	[0.0059]*
Constant	1.4696	1.1286	1.2191	1.2989	1.3634	1.4348	1.5100	1.5952	1.7211	1.9077
	[0.0091]***	[0.0104]***	[0.0114]***	[0.0107]***	[0.0107]***	[0.0109]***	[0.0120]***	[0.01387]***	[0.0157]***	[0.0179]*
Observations	25994	25994	25994	25994	25994	25994	25994	25994	25994	25994
R-squared	0.40	2000	20000	2000	2000	2000	2000	20771	20771	20000
					Specifi	ication 2				
Min. years of education	0.0166	0.0127	0.0131	0.0148	0.0158	0.0152	0.0158	0.0159	0.0175	0.0189
	[0.0007]***	[0.0008]***	[0.0008]***	[0.0007]***	[0.0008]***	[0.0008]***	[0.0007]***	[0.0008]***	[0.0010]***	[0.0014]*
Vocational degree	0.0614	0.0521	0.0549	0.0635	0.0668	0.0640	0.0668	0.0626	0.0633	0.0560
	[0.0063]***	[0.0077]***	[0.0076]***	[0.0063]***	[0.0073]***	[0.0070]***	[0.0069]***	[0.0074]***	[0.0093]***	[0.0127]*
Age:15-24 years old	-0.2713	-0.1503	-0.1824	-0.2004	-0.2185	-0.2355	-0.2704	-0.3059	-0.3418	-0.4120
	[0.0105]***	[0.0127]***	[0.0128]***	[0.0106]***	[0.0122]***	[0.0117]***	[0.0116]***	[0.0125]***	[0.0158]***	[0.0216]*
Age:25-34 years old	-0.1287	-0.0875	-0.0877	-0.1017	-0.1052	-0.1109	-0.1264	-0.1359	-0.1475	-0.1770
	[0.0070]***	[0.0082]***	[0.0084]***	[0.0070]***	[0.0081]***	[0.0078]***	[0.0078]***	[0.0084]***	[0.0106]***	[0.0145]*
Age:35-44 years old	-0.0016	0.0215	0.0188	0.0134	0.0075	0.0060	0.0032	-0.0016	-0.0101	-0.0240
	[0.0065]	[0.0078]***	[0.0078]**	[0.0065]**	[0.0075]	[0.0072]	[0.0071]	[0.0077]	[0.0096]	[0.0131]*
Age:45-54 years old	0.0452	0.0569	0.0576	0.0488	0.0477	0.0488	0.0466	0.0412	0.0332	0.0354
<u>.</u>	[0.0063]***	[0.0077]***	[0.0077]***	[0.0063]***	[0.0073]***	[0.0070]***	[0.0069]***	[0.0074]***	[0.0092]***	[0.0126]*
Tenure in years	0.0224	0.0237	0.0249	0.0247	0.0251	0.0244	0.0239	0.0218	0.0192	0.0152
<u>.</u>	[0.0008]***	[0.0010]***	[0.0010]***	[0.0008]***	[0.0009]***	[0.0009]***	[0.0009]***	[0.0009]***	[0.0012]***	[0.0016]*
Tenure squared/100	-0.0233	-0.0240	-0.0264	-0.0252	-0.0264	-0.0252	-0.0253	-0.0208	-0.0169	-0.0123
squared 100	[0.0026]***	[0.0033]***	[0.0033]***	[0.0027]***	[0.0031]***	[0.0029]***	[0.0029]***	[0.0031]***	[0.0039]***	[0.0054]*
Constant	1.6815	1.3497	1.3808	1.4119	1.5052	1.5677	1.7074	1.8968	2.0053	2.1575
Constant	[0.0582]***	[0.0697]***	[0.0696]***	[0.0576]***	[0.0663]***	[0.0635]***	[0.0626]***	[0.0674]***	[0.0842]***	[0.1157]*
Observations	25882	[0.0697]*** 25882	[0.0696]*** 25882	[0.0576]*** 25882	[0.0663]*** 25882	[0.0635]*** 25882	[0.0626]*** 25882	[0.0674]*** 25882	[0.0842]*** 25882	[0.1157]* 25882
Observations		23002	23002	23002	23002	23002	23002	23002	23002	23082
R-squared	0.53									



Table A4: 0	OLS an	l quantile	estimation	results,	Sample:	males,	Year: 2002	

	OLS				Quantile es					
		1	2	3	4	5	6	7	8	9
					Specif	fication 1				
Ain. years of education	0.0490	0.0345	0.0383	0.0417	0.0452	0.0464	0.0491	0.0515	0.0539	0.0579
	[0.0007]***	[0.0008]***	[0.0007]***	[0.0007]***	[0.0007]***	[0.0007]***	[0.0009]***	[0.0009]***	[0.0010]***	[0.0014]*
ocational degree	-0.0527	-0.0172	-0.0349	-0.0449	-0.0572	-0.0529	-0.0568	-0.0591	-0.0712	-0.0845
	[0.0067]***	[0.0086]**	[0.0080]***	[0.0078]***	[0.0076]***	[0.0076]***	[0.0089]***	[0.0086]***	[0.0091]***	[0.0126]*
Age:15-24 years old	-0.3965	-0.1837	-0.2237	-0.2564	-0.3091	-0.3652	-0.4267	-0.4783	-0.5557	-0.6599
	[0.0122]***	[0.0153]***	[0.0142]***	[0.0140]***	[0.0137]***	[0.0138]***	[0.0162]***	[0.0157]***	[0.0167]***	[0.0233]*
ge:25-34 years old	-0.2453	-0.1201	-0.1441	-0.1597	-0.1977	-0.2337	-0.2754	-0.2944	-0.3305	-0.3756
	[0.0095]***	[0.0117]***	[0.0109]***	[0.0108]***	[0.0106]***	[0.0106]***	[0.0125]***	[0.0126]***	[0.0130]***	[0.0182]*
ge:35-44 years old	-0.0740	-0.0286	-0.0267	-0.0215	-0.0419	-0.0649	-0.0829	-0.0894	-0.1168	-0.1405
	[0.0091]***	[0.0113]**	[0.0105]**	[0.0104]**	[0.0102]***	[0.0102]***	[0.0120]***	[0.0116]***	[0.0123]***	[0.0172]*
Age:45-54 years old	-0.0045	0.0497	0.0472	0.0431	0.0245	0.0038	-0.0144	-0.0280	-0.0518	-0.0641
	[0.0087]	[0.0109]***	[0.0101]***	[0.0100]***	[0.0097]**	[0.0097]	[0.0114]	[0.0110]**	[0.0117]***	[0.0163]*
enure in years	0.0285	0.0314	0.0311	0.0311	0.0315	0.0310	0.0299	0.0293	0.0274	0.0225
	[0.0008]***	[0.0011]***	[0.0010]***	[0.0010]***	[0.0009]***	[0.0009]***	[0.0011]***	[0.0010]***	[0.0011]***	[0.0016]*
enure squared/100	-0.0193	-0.0199	-0.0127	-0.0113	-0.0147	-0.0177	-0.0205	-0.0242	-0.0263	-0.0243
	[0.0029]***	[0.0040]***	[0.0036]***	[0.0034]***	[0.0033]***	[0.0033]***	[0.0038]***	[0.0037]***	[0.0039]***	[0.0055]*
onstant	1.3888	1.0507	1.1330	1.1909	1.2608	1.3579	1.4513	1.5417	1.6788	1.8715
	[0.0109]***	[0.0130]***	[0.0123]***	[0.0123]***	[0.0121]***	[0.0123]***	[0.0145]***	[0.0139]***	[0.0147]***	[0.0203]*
Observations	25964	25964	25964	25964	25964	25964	25964	25964	25964	25964
-squared	0.52									
						fication 2				
In. years of education	0.0208	0.0165	0.0168	0.0183	0.0184	0.0191	0.0195	0.0209	0.0217	0.0223
	[0.0008]***	[0.0011]***	[0.0009]***	[0.0009]***	[0.0009]***	[0.0010]***	[0.0009]***	[0.0010]***	[0.0012]***	[0.0015]*
ocational degree	-0.0124	-0.0123	-0.0146	-0.0116	-0.0128	-0.0095	-0.0104	-0.0184	-0.0239	-0.0202
	[0.0063]**	[0.0082]	[0.0067]**	[0.0067]*	[0.0069]*	[0.0073]	[0.0067]	[0.0074]**	[0.0092]***	[0.0116]*
ge:15-24 years old	-0.3346	-0.1822	-0.2342	-0.2486	-0.2839	-0.3012	-0.3179	-0.3507	-0.3895	-0.4229
	[0.0110]***	[0.0142]***	[0.0115]***	[0.0117]***	[0.0120]***	[0.0128]***	[0.0118]***	[0.0130***	[0.0163]***	[0.0208]
Age:25-34 years old	-0.2141	-0.1111	-0.1552	-0.1634	-0.1860	-0.2022	-0.2115	-0.2249	-0.2441	-0.2434
	[0.0085]***	[0.0108]***	[0.0088]***	[0.0090]***	[0.0092]***	[0.0099]***	[0.0091]***	[0.0101]***	[0.0126]***	[0.0162]*
ge:35-44 years old	-0.0685	-0.0254	-0.0470	-0.0473	-0.0578	-0.0618	-0.0637	-0.0718	-0.0760	-0.0621
	[0.0081]***	[0.0105]**	[0.0085]***	[0.0087]***	[0.0089]***	[0.0095]***	[0.0087]***	[0.0096]***	[0.0120]***	[0.0152]3
ge:45-54 years old	-0.0165	0.0215	0.0004	0.0052	-0.0048	-0.0129	-0.0202	-0.0241	-0.0289	-0.0189
	[0.0077]**	[0.0099]**	[0.0081]	[0.0082]	[0.0084]	[0.0090]	[0.0083]**	[0.0091]***	[0.0114]**	[0.0144]
enure in years	0.0216	0.0282	0.0267	0.0255	0.0242	0.0232	0.0227	0.0211	0.0192	0.0162
	[0.0007]***	[0.0010]***	[0.0008]***	[0.0008]***	[0.0008]***	[0.0009]***	[0.0008]***	[0.0009]***	[0.0011]***	[0.0014]*
enure squared/100	-0.0165	-0.0281	-0.0239	-0.0210	-0.0191	-0.0170	-0.0165	-0.0143	-0.0109	-0.0074
	[0.0026]***	[0.0034]***	[0.0028]***	[0.0028]***	[0.0028]***	[0.0030]***	[0.0028]***	[0.0031]***	[0.0039]***	[0.0049]
onstant	1.8026	1.3386	1.4204	1.5307	1.6210	1.8157	1.8786	1.9067	2.2022	2.2230
	[0.0490]***	[0.0578]***	[0.0495]***	[0.0512]***	[0.0528]***	[0.0565]***	[0.0520]***	[0.0567]***	[0.0658]***	[0.0828]*
Observations	25964	25964	25964	25964	25964	25964	25964	25964	25964	25964
R-squared	0.62									

Notes: as in Table A1.

In         2         3         4         5         6         7         8         9           Specification 1           Specification 1           Min. years of educatin         0.00601         0.00081***         0.00201***         0.00201***         0.00201***         0.00201***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0021***         0.0011***         0.011***         0.016**         0.016**         0.016**         0.016**         0.016***         0.016***         0.016***         0.016***         0.016***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.016****         0.017***         0.016****         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.017***         0.016****         0.017*** <td< th=""><th></th><th>OLS</th><th></th><th></th><th></th><th>Quantile esti</th><th>mations</th><th></th><th></th><th></th><th></th></td<>		OLS				Quantile esti	mations				
Min. years of education         0.0360         0.0161         0.0204         0.0242         0.0273         0.0320         0.0346         0.0376         0.0428         0.02           Vocational degree         0.1027         0.0942         0.0087**         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00081***         0.00221***         0.00081***         0.01181**         0.01181**         0.01281***         0.0161***         0.0171***         0.01071***         0.01071***         0.01071***         0.00171***         0.0171***         0			1	2	3	4	5	6	7	8	9
000071***         000081***         000081***         000081***         000081***         000081***         000081***         000081***         000081***         000081***         000081***         000081***         000081***         000081***         000081***         000081***         000071***         00071**         00071***         00071**         000771**         00071**         00071**						Specifi	cation 1				
Vacational degree (0.027)0.027 (0.02301)***0.027 (0.02301)***0.0181 (0.02301***0.0870 (0.02301***0.0870 (0.02301***0.0870 (0.02301***0.0870 (0.02301***0.0870 (0.02301***0.0131 (0.0131***0.0161***0.0131**0.0125***0.00251***0.00251***0.00251***0.002781***0.002781***0.002781***0.002781***0.002781***0.002781***0.002781***0.002781***0.0021***0.0021***0.00151***0.0131***0.0151***0.0151***0.0171***0.	Min. years of education	0.0360		0.0204	0.0242	0.0273	0.0320	0.0346	0.0376	0.0428	0.0501
Instruct		[0.0007]***	[0.0008]***	[0.0005]***	[0.0006]***	[0.0008]***	[0.0008]***		[0.0010]***	[0.0011]***	[0.0019]*
Age: 15-24 years old         0.1971         -0.0843         -0.1041         -0.1259         -0.1594         -0.1830         -0.1780         -0.2000         -0.2187         -0.3           Age: 25-34 years old         -0.0131         -0.0151         -0.0277         -0.0449         -0.0373         -0.0654         -0.057         -0.0661         0.02         -0.0673         0.0667         0.0730         0.0673         0.0255         0.0274	Vocational degree	0.1027	0.0942	0.0878	0.1216	0.1121	0.1057	0.1181	0.1096	0.0870	0.0808
Image: 1.3.1 mode:		[0.0210]***	[0.0236]***	[0.0166]***	[0.0189]***	[0.0225]***	[0.0242]***	[0.0256]***	[0.0278]***	[0.0286]***	[0.0470]*
Age: 25-34 years old Age: 35-44 years old Age: 35-44 years old Age: 35-44 years old 0.0127)***         0.01421 (0.0127)***         0.01421 (0.0127)***         0.01421 (0.0127)***         0.01421 (0.0127)***         0.01421 (0.017)***         0.01431 (0.017)***         0.01431 (0.017)***         0.01431***         0.0050 (0.012)***         0.0173***         0.0050 (0.012)***         0.01631***         0.01601 (0.012)***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.01631***         0.0173***         0.0173***         0.011***         0.011***         0.011***         0.011***         0.011***         0.011***         0.011***         0.011***         0.011***         0.011***         0.011***         0.011***         0.011***         0.011***         0.001*	Age:15-24 years old	-0.1971	-0.0843	-0.1041	-0.1259	-0.1594	-0.1830	-0.1780	-0.2000	-0.2187	-0.3010
Image: 1         [0.01278]***         [0.0142]         [0.0112]         [0.0113]**         [0.0137]***         [0.0171]***         [0.0171]***         [0.0171]***         [0.0178]***         [0		[0.0143]***	[0.0161]***	[0.0115]***	[0.0128]***	[0.0154]***	[0.0166]***	[0.0177]***	[0.0193]***	[0.0202]***	[0.0335]*
Age:35-44         years old         0.0526         0.0673         0.0676         0.0737         0.0660         0.0609         0.0739         0.0673         0.0661         0.02           Age:45-54         years old         0.0522         0.0373         0.0582         0.0668         0.0529         0.01631***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0152]***         [0.0151]***         [0.0151]***         [0.0151]***         [0.0151]***         [0.0051]**         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0051]***         [0.0151]***         [0.0151]***         [0.0151]***         [0.0151]***         [0.016]****         [0.0151]***         [0.0151]***         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]*****         [0.016]*****         [	Age:25-34 years old	-0.0580	-0.0140	-0.0131	-0.0216	-0.0297	-0.0449	-0.0355	-0.0491	-0.0545	-0.0949
Image: 10.0122]***         [0.0138]***         [0.0099]***         [0.0110]***         [0.0122]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0172]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0164]***         [0.015]***         [0.015]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.015]***         [0.015]***         [0.015]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0015]***         [0.0015]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.00163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163]***         [0.0163		[0.01278]***	[0.0142]	[0.0102]	[0.0114]*	[0.0137]**	[0.0147]***	[0.0157]**	[0.0171]***	[0.0178]***	[0.0295]*
Age:45-54 years old [0.012]***         0.052         0.0373         0.0582         0.0668         0.0589         0.0660         0.0710         0.0637         0.073         0.073           Fenure in years         0.00245         0.0236         0.0243         0.0255         0.0244         0.0246         0.0247         0.015]***         [0.001]***         [0.016]***         [0.016]***         [0.016]***         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]****         [0.016]*****         [0.016]*****         [0.016]*****         [0.016]***** </td <td>Age:35-44 years old</td> <td>0.0526</td> <td>0.0525</td> <td>0.0676</td> <td>0.0737</td> <td>0.0660</td> <td>0.0609</td> <td>0.0739</td> <td>0.0673</td> <td>0.0661</td> <td>0.0259</td>	Age:35-44 years old	0.0526	0.0525	0.0676	0.0737	0.0660	0.0609	0.0739	0.0673	0.0661	0.0259
Image: Section of the sectio		[0.0122]***	[0.0138]***	[0.0099]***	[0.0110]***	[0.0132]***	[0.0142]***	[0.0150]***	[0.0163]***	[0.0169]***	[0.0278]
Tenure in years         0.0245         0.0236         0.0243         0.0255         0.0244         0.0246         0.0247         0.0257         0.0256         0.0267           Tenure squared/100         -0.0149 $[0.0011]^{***}$ $[0.0005]^{***}$ $[0.0005]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0015]^{***}$ $[0.0051]^{***}$ $[0.0051]^{***}$ $[0.0051]^{***}$ $[0.0051]^{***}$ $[0.0051]^{***}$ $[0.0053]^{***}$ $[0.0041]^{***}$ $[0.0041]^{***}$ $[0.0041]^{***}$ $[0.0041]^{***}$ $[0.0041]^{***}$ $[0.0041]^{***}$ $[0.0169]^{***}$ $[0.018]^{***}$ $[0.011]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0111]^{***}$ $[0.0$	Age:45-54 years old	0.0562	0.0373	0.0582	0.0668	0.0589	0.0660	0.0710	0.0637	0.0673	0.0380
Tenure in years         0.0245         0.0236         0.0243         0.0255         0.0244         0.0246         0.0247         0.0257         0.0256         0.001           Tenure squared/10         0.0011***         [0.0012]***         [0.00013***         [0.00013***         [0.00013***         [0.0013***         [0.0015***         [0.0015***         [0.0015***         [0.0015***         [0.0015***         [0.0044]***         [0.0044]***         [0.0044]***         [0.0041***         [0.0141***         [0.0141***         [0.011***         [0.011***         [0.011***         [0.011***         [0.011***         [0.011***         [0.011***         [0.0001***         [0.0001***         [0.0001***         [0.0001***         [0.011***         [0.011***         [0.0001***         [0.0001***         [0.0001***         [0.0001***         [0.0001***         [0.0001***         [0.0001***         [0.0001***	-	[0.0125]***	[0.0142]***	[0.0101]***	[0.0113]***	[0.0135]***	[0.0145]***	[0.0154]***	[0.0166]***	[0.0172]***	[0.0282]
Image: Bold II (0.0012)***         Image: Bold II (0.0011)***         Image: Bold II (0.0011)*** <th< td=""><td>Tenure in years</td><td>0.0245</td><td>0.0236</td><td>0.0243</td><td>0.0255</td><td>0.0244</td><td>0.0246</td><td>0.0247</td><td>0.0257</td><td>0.0256</td><td>0.0221</td></th<>	Tenure in years	0.0245	0.0236	0.0243	0.0255	0.0244	0.0246	0.0247	0.0257	0.0256	0.0221
Tenure squared/100-0.0149-0.0217-0.0143-0.0142-0.0079-0.0070-0.0066-0.0166-0.0173-0.0070Constant[0.0051]***[0.0051]***[0.0036]***[0.004]***[0.0051]**[0.0051]***[0.0058]**[0.0058]**[0.0058]**[0.0057]***[0.0058]**[0.0058]**[0.0057]***[0.0159]***[0.0169]***[0.0169]***[0.0159]***[0.011]***[0.011]***[0.011]***[0.0009]***[0.011]***[0.011]***[0.0019]***[0.0019]***[0.011]***[0.011]***[0.011]***[0.011]***[0.011]***[0.011]***[0.011]***[0.011]***[0.011]***[0.011]***[0.011]***[0.012]***[0.01		[0.0011]***	[0.0012]***	[0.0009]***	[0.0010]***	[0.0012]***	[0.0013]***	[0.0013]***		[0.0015]***	[0.0024]*
Image: biologeneration         Image: biologeneration <thimage: biologeneration<="" th="">         Image: biologeneration<!--</td--><td>Tenure squared/100</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>-0.0134</td></thimage:>	Tenure squared/100										-0.0134
Constant         1.1761         1.117         1.169         1.1260         1.1588         1.1708         1.1913         1.2325         1.2706         1.38           Observations         12077	1										[0.0095]
10.0136}***         10.0147}***         10.0157***         10.0169****         10.0184****         10.017***         10.017***         10.017***         10.017***         10.017***         10.011***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121***         10.0121*** </td <td>Constant</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>1.3881</td>	Constant										1.3881
Deservations R-squared         12077         1207           100121 <td>constant</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>[0.0308]*</td>	constant										[0.0308]*
R-squared         0.43           Septence         Specification 2           Min. years of eduction         0.0151         0.0089         0.0013         0.0112         0.0125         0.0138         0.0148         0.0149         0.001           Yocational degree         0.0013         0.0011         0.0021         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121         0.0121 </td <td>Observations</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>12077</td>	Observations										12077
Specification 2           Min. years of education         0.0151         0.0089         0.0093         0.0103         0.0112         0.0125         0.0138         0.0148         0.0159         0.00           Vocational degree         0.0681         0.0295         0.0226         0.0518         0.00171***         [0.0011]***         [0.0011]***         [0.0020]***         [0.0201]***         [0.0208]** <t< td=""><td></td><td></td><td>12077</td><td>12077</td><td>12077</td><td>12077</td><td>12077</td><td>12077</td><td>12077</td><td>12077</td><td>12077</td></t<>			12077	12077	12077	12077	12077	12077	12077	12077	12077
Min. years of education $0.0151$ $0.0089$ $0.0093$ $0.0103$ $0.0112$ $0.0125$ $0.0138$ $0.0148$ $0.0159$ $0.01$ Vocational degree $0.0681$ $0.0295$ $0.0226$ $0.0518$ $0.0624$ $0.0771$ $0.0876$ $0.0896$ $0.0586$ $0.096$ $0.016]^{***}$ $[0.0012]^{***}$ $[0.0212]$ $[0.0211]$ $[0.0220]^{**}$ $[0.0175]^{***}$ $[0.0214]^{***}$ $[0.024]^{***}$ $[0.0242]^{***}$ $[0.0288]^{***}$ $[0.0211]^{***}$ $[0.0128]^{***}$ $[0.0142]^{***}$ $[0.0128]^{***}$ $[0.0168]^{***}$ $[0.0211]^{***}$ $[0.0221]^{***}$ $[0.0128]^{***}$ $[0.0168]^{***}$ $[0.0188]^{***}$ $[0.0188]^{***}$ $[0.0188]^{***}$ $[0.0188]^{***}$ $[0.0188]^{***}$ $[0.0188]^{***}$ $[0.017$	R-squared	0.45									
$(0.0010)^{**}$ $(0.0012]^{**}$ $(0.0011)^{**}$ $(0.0012]^{**}$ $(0.0011)^{**}$ $(0.0011)^{**}$ $(0.0013)^{**}$ $(0.0016)^{**}$ $(0.0022)^{**}$ $(0.017)^{**}$ $(0.0121)^{**}$ $(0.011)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.0121)^{**}$ $(0.011)^{**}$ $(0.011)^{**}$ $(0.011)^{**}$ $(0.011)^{**}$ $(0.011)^{**}$ $(0.011)^{**}$ <						Specifi	cation 2				
Vocational degree $0.0681$ $0.0295$ $0.0226$ $0.0518$ $0.0624$ $0.0771$ $0.0876$ $0.0896$ $0.0586$ $0.0586$ $0.0586$ $Age:15-24$ years old $-0.1493$ $-0.0914$ $-0.0913$ $-0.1116$ $-0.1193$ $-0.1295$ $-0.1337$ $-0.1382$ $-0.1490$ $-0.2$ $[0.0127]^{***}$ $[0.0145]^{***}$ $[0.015]^{***}$ $[0.01001^{**}$ $[0.0142]^{***}$ $[0.0169]^{***}$ $[0.0201]^{***}$ $[0.0201]^{***}$ $Age:25-34$ years old $-0.0405$ $-0.0077$ $-0.0056$ $-0.0143$ $-0.202$ $-0.293$ $-0.360$ $-0.388$ $-0.0406$ $-0.00$ $[0.0113]^{***}$ $[0.0129]$ $[0.0128]$ $[0.0133]$ $[0.0107]^{**}$ $[0.0131]^{**}$ $[0.0142]^{***}$ $[0.0178]^{***}$ $[0.0178]^{***}$ $[0.0188]^{***}$ $[0.0188]^{***}$ $[0.0169]^{***}$ $[0.0181]^{***}$ $[0.0181]^{***}$ $[0.0168]^{***}$ $[0.0181]^{***}$ $[0.0181]^{***}$ $[0.0178]^{***}$ $[0.0183]^{***}$ $[0.0183]^{***}$ $[0.0182]^{***}$ $[0.0126]^{***}$ $[0.0142]^{***}$ $[0.0188]^{***}$ $[0.0188]^{***}$ $[0.0129]^{***}$ $[0.0125]^{***}$ $[0.0126]^{***}$ $[0.0142]^{***}$ $[0.0178]^{***}$ $[0.0178]^{***}$ $[0.0168]^{***}$ $[0.0188]^{***}$ $[0.0188]^{***}$ $[0.0188]^{***}$ $[0.0129]^{***}$ $[0.0129]^{***}$ $[0.0129]^{***}$ $[0.0142]^{***}$ $[0.0179]^{***}$ $[0.012]^{***}$ $[0.0142]^{***}$ $[0.0170]^{***}$ $[0.0168]^{***}$ $[0.0168]^{***}$ $[0.0129]^{***}$ $[0.0129]^{***}$ $[0.0129]^{***}$ <	Min. years of education	0.0151	0.0089	0.0093	0.0103	0.0112	0.0125	0.0138	0.0148	0.0159	0.0183
$(0.0186)^{***}$ $(0.0212]$ $(0.0211]$ $(0.0220)^{***}$ $(0.0175)^{***}$ $(0.0214)^{***}$ $(0.024)^{***}$ $(0.024)^{***}$ $(0.0288)^{**}$ $(0.0288)^{**}$ $(0.0288)^{**}$ $(0.0288)^{**}$ $(0.0288)^{***}$ $(0.0218)^{***}$ $(0.012)^{***}$ $(0.012)^{***}$ $(0.0168)^{***}$ $(0.0201)^{***}$ $(0.0178)^{***}$ $(0.0178)^{***}$ $(0.018)^{***}$ $(0.012)^{***}$ $(0.0168)^{***}$ $(0.0188)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$ $(0.0168)^{***}$		[0.0010]***	[0.0012]***	[0.0011]***	[0.0012]***	[0.0009]***	[0.0011]***	[0.0011]***	[0.0013]***	[0.0016]***	[0.0019]*:
$0.0186^{+**}$ $0.0212$ $0.0211$ $0.020^{+**}$ $0.0175^{+**}$ $0.0214^{+**}$ $0.024^{+**}$ $0.0242^{+**}$ $0.0288^{+*}$ $0.0188^{+*}$ $0.0188^{+*}$ $0.0188^{+*}$ $0.0188^{+*}$ $0.0188^{+*}$ $0.0188^{+*}$ $0.0188^{+*}$ $0.0188^{+*}$ $0.0188^{+*}$ $0.0188^{+*}$ $0.0188^{+*$	Vocational degree	0.0681	0.0295	0.0226	0.0518	0.0624	0.0771	0.0876	0.0896	0.0586	0.0962
$ \begin{bmatrix} 0.0127 \end{bmatrix}^{**} & \begin{bmatrix} 0.0145 \end{bmatrix}^{***} & \begin{bmatrix} 0.0155 \end{bmatrix}^{***} & \begin{bmatrix} 0.0150 \end{bmatrix}^{***} & \begin{bmatrix} 0.0120 \end{bmatrix}^{***} & \begin{bmatrix} 0.0147 \end{bmatrix}^{***} & \begin{bmatrix} 0.0142 \end{bmatrix}^{***} & \begin{bmatrix} 0.0169 \end{bmatrix}^{***} & \begin{bmatrix} 0.0201 \end{bmatrix}^{***} & \begin{bmatrix} 0.0169 \end{bmatrix}^{***} & \begin{bmatrix} 0.0201 \end{bmatrix}^$	-	[0.0186]***	[0.0212]	[0.0211]	[0.0220]**	[0.0175]***	[0.0214]***	[0.0204]***	[0.0242]***	[0.0288]**	[0.0353]*
$ \begin{bmatrix} 0.0127 \end{bmatrix}^{**} & \begin{bmatrix} 0.0145 \end{bmatrix}^{***} & \begin{bmatrix} 0.0155 \end{bmatrix}^{***} & \begin{bmatrix} 0.0150 \end{bmatrix}^{***} & \begin{bmatrix} 0.0120 \end{bmatrix}^{***} & \begin{bmatrix} 0.0147 \end{bmatrix}^{***} & \begin{bmatrix} 0.0142 \end{bmatrix}^{***} & \begin{bmatrix} 0.0169 \end{bmatrix}^{***} & \begin{bmatrix} 0.0201 \end{bmatrix}^{***} & \begin{bmatrix} 0.0169 \end{bmatrix}^{***} & \begin{bmatrix} 0.0201 \end{bmatrix}^$	Age:15-24 years old										-0.2126
Age:25-34 years old $-0.0405$ $-0.0077$ $-0.0056$ $-0.0143$ $-0.0202$ $-0.0293$ $-0.0360$ $-0.0388$ $-0.0406$ $-0.0$ Age:35-44 years old $0.0524$ $0.0648$ $0.0667$ $0.0568$ $0.0561$ $0.0544$ $0.0503$ $0.0479$ $0.0486$ $0.01$ Age:45-54 years old $0.0480$ $0.0430$ $0.0520$ $0.0482$ $0.0511$ $0.0471$ $0.0405$ $0.0464$ $0.0468$ $0.01$ Age:45-54 years old $0.0480$ $0.0430$ $0.0520$ $0.0482$ $0.0511$ $0.0471$ $0.0405$ $0.0464$ $0.0468$ $0.01$ Ion10]*** $[0.0127]^{***}$ $[0.0127]^{***}$ $[0.0123]^{***}$ $[0.0127]^{***}$ $[0.0122]^{***}$ $[0.0144]^{***}$ $[0.0170]^{***}$ Fenure in years $0.0188$ $0.0239$ $0.0212$ $0.0195$ $0.0191$ $0.0182$ $0.0176$ $0.0172$ $0.0152$ $0.01$ Fenure squared/100 $-0.0063$ $-0.0323$ $-0.0162$ $-0.0060$ $-0.0056$ $-0.0099$ $0.0023$ $0.0025$ $0.0118$ $0.01$ Formure squared/100 $-0.0063$ $-0.0323$ $-0.0162$ $-0.0060$ $-0.0056$ $-0.0099$ $0.0023$ $0.0025$ $0.0118$ $0.011$ Constant $1.4723$ $1.3263$ $1.3571$ $1.3595$ $1.4001$ $1.4654$ $1.5310$ $1.5905$ $1.6070$ $1.77$ $[0.039]^{***}$ $[0.042]^{***}$ $[0.044]^{***}$ $[0.0369]^{***}$ $[0.047]^{***}$ $[0.049]^{***}$ $[0.060]^{***}$ <td>0</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>[0.0251]*</td>	0										[0.0251]*
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Age:25-34 years old										-0.0748
Age:35-44 years old         0.0524         0.0648         0.0667         0.0568         0.0561         0.0544         0.0503         0.0479         0.0486         0.01           Age:45-54 years old         0.0480         0.0430         0.0520         0.0482         0.0511         0.012]***         [0.010]***         [0.012]***         [0.010]***         [0.012]***         [0.011]***         [0.01											[0.0219]*
$ \begin{bmatrix} 0.0108 \end{bmatrix}^{***} & \begin{bmatrix} 0.0124 \end{bmatrix}^{***} & \begin{bmatrix} 0.0124 \end{bmatrix}^{***} & \begin{bmatrix} 0.0129 \end{bmatrix}^{***} & \begin{bmatrix} 0.0103 \end{bmatrix}^{***} & \begin{bmatrix} 0.0125 \end{bmatrix}^{***} & \begin{bmatrix} 0.0120 \end{bmatrix}^{***} & \begin{bmatrix} 0.0124 \end{bmatrix}^{***} & \begin{bmatrix} 0.0168 \end{bmatrix}^{***} & \begin{bmatrix} 0.012 \end{bmatrix}^{***} & \begin{bmatrix} 0.0112 \end{bmatrix}^{***} & \begin{bmatrix} 0.0011 \end{bmatrix}^{***}$	Age:35-44 years old										0.0133
Age:45-54 years old         0.0480         0.0430         0.0520         0.0482         0.0511         0.0471         0.0405         0.0464         0.0468         0.01           Io.01101***         [0.0127]***         [0.0127]***         [0.0132]***         [0.015]***         [0.0127]***         [0.0127]***         [0.0127]***         [0.0127]***         [0.0127]***         [0.012]***         [0.017]****         [0.017]****         [0.017]***	ingense in years ora										[0.0204]
[0.0110]***         [0.0127]***         [0.0127]***         [0.0123]***         [0.0105]***         [0.0127]***         [0.0127]***         [0.0170]***         [0.0172]***	Age: 45-54 years old										0.0183
Tenure in years         0.0188         0.0239         0.0212         0.0195         0.0191         0.0182         0.0176         0.0172         0.0152         0.01           Iono101***         [0.0011]***         [0.0011]***         [0.0012]***         [0.0009]***         [0.0011]***         [0.0011]***         [0.0011]***         [0.0011]***         [0.0011]***         [0.0013]***         [0.0015]***         [0.0015]***         [0.0015]***         [0.0015]***         [0.0011]***         [0.0011]***         [0.0013]***         [0.0015]***	Age. 45-54 years old										[0.0206]
[0.0010]***         [0.0011]***         [0.0011]***         [0.0012]***         [0.0009]***         [0.0011]***         [0.0011]***         [0.0013]***         [0.0015]***	Tonura in voora										
Tenure squared/100         -0.0063         -0.0323         -0.0162         -0.0060         -0.0056         -0.0009         0.0023         0.0025         0.0118         0.01           [0.0040]         [0.0045]***         [0.0044]***         [0.0047]         [0.0037]         [0.0046]         [0.0043]         [0.0051]         [0.0060]**         [0.0           Constant         1.4723         1.3263         1.3571         1.3595         1.4001         1.4654         1.5310         1.5905         1.6070         1.77           [0.0390]***         [0.042]***         [0.045]***         [0.0458]***         [0.0369]***         [0.0447]***	renute in years										
$ \begin{bmatrix} 0.0040 \end{bmatrix} \begin{bmatrix} 0.0045 \end{bmatrix}^{***} \begin{bmatrix} 0.0044 \end{bmatrix}^{***} \begin{bmatrix} 0.0047 \end{bmatrix} \begin{bmatrix} 0.0037 \end{bmatrix} \begin{bmatrix} 0.0037 \end{bmatrix} \begin{bmatrix} 0.0046 \end{bmatrix} \begin{bmatrix} 0.0043 \end{bmatrix} \begin{bmatrix} 0.0051 \end{bmatrix} \begin{bmatrix} 0.0060 \end{bmatrix}^{**} \begin{bmatrix} 0.006 \end{bmatrix}^{**} \\ 1.4723 \end{bmatrix} \begin{bmatrix} 1.3263 \\ 1.3571 \\ 1.3595 \\ 1.4001 \end{bmatrix} \begin{bmatrix} 1.4654 \\ 1.5310 \\ 1.5905 \\ 1.6070 \\ 1.77 \\ 1.0399 \end{bmatrix}^{***} \begin{bmatrix} 0.0424 \end{bmatrix}^{***} \begin{bmatrix} 0.0446 \end{bmatrix}^{***} \begin{bmatrix} 0.0458 \end{bmatrix}^{***} \begin{bmatrix} 0.0369 \end{bmatrix}^{***} \begin{bmatrix} 0.0447 \end{bmatrix}^{***} \begin{bmatrix} 0.0427 \end{bmatrix}^{***} \begin{bmatrix} 0.0499 \end{bmatrix}^{***} \begin{bmatrix} 0.060 \end{bmatrix}^{**} \\ 1.570 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 \\ 1.670 \\ 1.77 $	T										[0.0019]*
Constant $1.4723$ $1.3263$ $1.3571$ $1.3595$ $1.4001$ $1.4654$ $1.5310$ $1.5905$ $1.6070$ $1.77$ $[0.0390]^{***}$ $[0.0424]^{***}$ $[0.0446]^{***}$ $[0.0369]^{***}$ $[0.0447]^{***}$ $[0.0499]^{***}$ $[0.0600]^{***}$ $[0.0600]^{***}$	renure squared/100										0.0165
$[0.0390]^{***}$ $[0.0424]^{***}$ $[0.0446]^{***}$ $[0.0458]^{***}$ $[0.0369]^{***}$ $[0.0447]^{***}$ $[0.0427]^{***}$ $[0.0499]^{***}$ $[0.0600]^{***}$ $[0.0$											[0.0074]*
	Constant										1.7782
Observations 12019 12019 12019 12019 12019 12019 12019 12019 12019 12019 12019 12019 12019 12019 12019											[0.0697]*
R-squared 0.57	Observations		12019	12019	12019	12019	12019	12019	12019	12019	12019

## Table A5: OLS and quantile estimation results, Sample: females, Year: 1995

	OLS				Quantile est	imations				
	OLD	1	2	3	4	5	6	7	8	9
					Specif	ication 1				
Min. years of education	0.0497	0.0307	0.0341	0.0380	0.0430	0.0455	0.0504	0.0517	0.0549	0.0606
	[0.0009]***	[0.0009]***	[0.0008]***	[0.00071]***	[0.0009]***	[0.0011]***	[0.0011]***	[0.0012]***	[0.0015]***	[0.0017]**
Vocational degree	-0.0225	0.0121	-0.0148	-0.0314	-0.0346	-0.0300	-0.0348	-0.0401	-0.0353	0.0061
	[0.0111]**	[0.0133]	[0.0113]	[0.00991]***	[0.0113]***	[0.0131]**	[0.0138]**	[0.0134]***	[0.0167]**	[0.0185]
Age:15-24 years old	-0.2834	-0.1571	-0.1660	-0.1977	-0.2599	-0.2895	-0.3239	-0.3359	-0.3476	-0.4266
	[0.0161]***	[0.0193]***	[0.0167]***	[0.01421]***	[0.0164]***	[0.0189]***	[0.0200]***	[0.0195]***	[0.0245]***	[0.0273]**
Age:25-34 years old	-0.1613	-0.0804	-0.0894	-0.1108	-0.1516	-0.1592	-0.1935	-0.1920	-0.1921	-0.2282
	[0.0145]***	[0.0175]***	[0.0147]***	[0.0129]***	[0.0148]***	[0.0171]***	[0.0180]***	[0.0176]***	[0.0220]***	[0.0243]**
Age:35-44 years old	-0.0507	-0.0151	-0.0123	-0.0150	-0.0362	-0.0360	-0.0650	-0.0622	-0.0631	-0.0995
45 54 11	[0.0142]***	[0.0171]	[0.0144]	[0.0126]	[0.0144]**	[0.0167]**	[0.0175]***	[0.0171]***	[0.0213]***	[0.0235]**
Age:45-54 years old	-0.0148	0.0078	0.0149	0.0156	-0.0140	-0.0149	-0.0381 [0.0178]**	-0.0281	-0.0141	-0.0419
Forum in voors	[0.01448] 0.0313	[0.0172] 0.0328	[0.0146] 0.0302	[0.0128] 0.0286	[0.0147] 0.0281	[0.0170] 0.0283	0.0288	[0.0173] 0.0296	[0.0217] 0.0281	[0.0239]* 0.0267
Fenure in years	[0.0010]***	[0.0012]***	[0.0011]***	[0.0009]***	[0.0281][0.0011]***	[0.0283	[0.0288	[0.0290	[0.0281	[0.0017]**
Tenure squared/100	-0.0251	-0.0406	-0.0247	-0.0112	-0.0029	-0.0023	-0.0042	-0.0112	-0.0114	-0.0171
renare squared/100	[0.0043]***	[0.0053]***	[0.00461]***	[0.0040]***	[0.0045]	[0.0051]	[0.0052]	[0.0051]**	[0.0064]*	[0.0073]**
Constant	1.1660	0.9943	1.0553	1.0959	1.1374	1.1821	1.2264	1.2936	1.3615	1.4854
constant	[0.0155]***	[0.0174]***	[0.01481]***	[0.0132]***	[0.0155]***	[0.0182]***	[0.0195]***	[0.0192]***	[0.0242]***	[0.0267]**
Observations	15485	15485	15485	15485	15485	15485	15485	15485	15485	15485
R-squared	0.45									
-										
	0.0154	0.0100		0.01.00		ication 2	0.0454	0.0144	0.0100	0.0405
Min. years of education		0.0129	0.0144	0.0138	0.0142	0.0146	0.0156	0.0164	0.0180	0.0197
7	[0.0011]***	[0.0014]***	[0.0011]***	[0.0010]***	[0.0010]***	[0.0009]***	[0.0011]***	[0.0013]***	[0.0015]***	[0.0021]**
Vocational degree	-0.0109	0.0097	-0.0022	-0.0041	-0.0166	-0.0198	-0.0153	-0.0204	-0.0154	-0.0090
Agails 24 years old	[0.0097] -0.2420	[0.0123] -0.1265	[0.0094] -0.1400	[0.0090] -0.1728	[0.0087]* -0.1896	[0.0082]** -0.1957	[0.0100] -0.2226	[0.0118]* -0.2569	[0.0125] -0.2787	[0.0181] -0.3359
Age:15-24 years old	-0.2420 [0.0140]***	-0.1265 [0.0180]***	-0.1400 [0.0136]***	-0.1728 [0.0131]***	-0.1896 [0.0126]***	-0.1957 [0.0119]***	-0.2226 [0.0146]***	-0.2569 [0.0173]***	-0.2787 [0.0187]***	-0.3339 [0.0270]**
Age:25-34 years old	-0.1348	-0.0605	-0.0742	-0.0982	-0.1081	-0.1062	-0.1200	-0.1445	-0.1479	-0.1683
Age.25-54 years old	[0.0126]***	[0.0162]***	[0.0121]***	[0.0117]***	[0.0113]***	[0.0106]***	[0.0131]***	[0.0155]***	[0.0167]***	[0.0239]**
Age:35-44 years old	-0.0432	0.0082	-0.0011	-0.0196	-0.0302	-0.0279	-0.0346	-0.0534	-0.0480	-0.0545
igenee it years ond	[0.0122]***	[0.0158]	[0.0119]	[0.0114]*	[0.0110]***	[0.0103]***	[0.0127]***	[0.0150]***	[0.0161]***	[0.0229]**
Age:45-54 years old	-0.0135	0.0182	0.0115	-0.0029	-0.0160	-0.0044	-0.0108	-0.0275	-0.0183	-0.0229
-8	[0.0124]	[0.0159]	[0.0120]	[0.0115]	[0.0112]	[0.0105]	[0.0129]	[0.0152]*	[0.0163]	[0.0232]
Fenure in years	0.0251	0.0325	0.0284	0.0248	0.0236	0.0233	0.0224	0.0208	0.0191	0.0171
2	[0.0009]***	[0.0012]***	[0.0009]***	[0.0009]***	[0.0008]***	[0.0008]***	[0.0009]***	[0.0011]***	[0.0012]***	[0.0017]**
Fenure squared/100	-0.0263	-0.0526	-0.0351	-0.0208	-0.0154	-0.0134	-0.0106	-0.0062	-0.0043	-0.0034
-	[0.0038]***	[0.0047]***	[0.0036]***	[0.0035]***	[0.0034]***	[0.0032]***	[0.0039]***	[0.0046]	[0.0049]	[0.0072]
Constant	1.6458	1.3300	1.4183	1.5260	1.5788	1.5813	1.6084	1.7160	1.9706	2.0660
	[0.0436]***	[0.0546]***	[0.0415]***	[0.0399]***	[0.0390]***	[0.0367]***	[0.0451]***	[0.0525]***	[0.0560]***	[0.0807]**
Observations	15485	15485	15485	15485	15485	15485	15485	15485	15485	15485
R-squared	0.6									

Table A6: OLS and quantile estimation results, Sample: females, Year: 2002

Notes: as in Table A1.

	1	2	3	4	5	6	7	8	9
					All				
Observed pay change (in logs)	-0.021	-0.014	-0.008	-0.008	0.007	0.048	0.100	0.147	0.195
Composition effects of:									
worker characteristics	-0.049	-0.056	-0.062	-0.068	-0.083	-0.059	-0.063	-0.035	0.03
job characteristics	-0.033	-0.012	-0.017	-0.026	-0.010	0.007	0.062	0.094	0.11
Price effects of:									
constant	-0.001	0.044	0.092	0.154	0.127	0.126	0.095	0.278	0.177
worker characteristics	0.036	0.011	-0.002	-0.018	-0.034	-0.028	-0.020	-0.014	0.006
job characteristics	0.000	-0.014	-0.039	-0.072	-0.025	-0.015	0.020	-0.180	-0.094
Residual effects	0.026	0.012	0.020	0.021	0.031	0.017	0.005	0.003	-0.03
					Males				
Observed pay change (in logs)	-0.046	-0.041	-0.030	-0.013	0.027	0.084	0.124	0.174	0.21
Composition effects of:									
worker characteristics	-0.068	-0.072	-0.073	-0.057	-0.081	-0.041	-0.023	0.027	0.06
job characteristics	-0.039	-0.020	-0.018	-0.010	0.001	0.076	0.077	0.112	0.11
Price effects of:									
constant	-0.011	0.040	0.119	0.116	0.248	0.171	0.010	0.197	0.06
worker characteristics	0.023	-0.016	-0.012	-0.041	-0.032	-0.032	-0.016	-0.005	0.01
job characteristics	0.024	0.010	-0.055	-0.021	-0.157	-0.095	0.068	-0.141	-0.04
Residual effects	0.025	0.017	0.009	0.000	0.048	0.005	0.008	-0.015	-0.004
Observed and shares (in 1 )	0.004	0.024	0.027	0.042	Females	0.074	0.100	0.460	0.047
Observed pay change (in logs)	0.001	0.021	0.037	0.043	0.060	0.074	0.109	0.163	0.213
Composition effects of:	0.040	0.027	0.025	0.044	0.045	0.024	0.020	0.007	0.00
worker characteristics	-0.042	-0.037	-0.035	-0.041	-0.045	-0.031	-0.028	0.007	0.036
job characteristics	-0.024	-0.014	-0.008	-0.013	0.002	0.016	0.013	0.061	0.08
Price effects of:	0.004	0.064	0.460	0.470	0.110	0.077	0.405	0.264	0.000
constant	0.004	0.061	0.166	0.179	0.116	0.077	0.125	0.364	0.28
worker characteristics	0.023	0.016	-0.015	-0.026	-0.020	-0.029	-0.055	-0.052	-0.03
job characteristics	-0.004	-0.024	-0.085	-0.070	-0.007	0.048	0.042	-0.195	-0.12
Residual effects	0.044	0.019	0.013	0.014	0.014	-0.008	0.011	-0.022	-0.03

## Table A7: Breakdown of observed wage changed by decile



Figure A1: Bootstrapped employee characteristics by sex and year



